Point process models for extreme returns: Harnessing implied volatility

Abstract

Forecasting the risk of extreme losses is an important issue in the management of financial risk. There has been a great deal of research examining how option implied volatilities (IV) can be used to forecast asset return volatility. However, the role of IV in the context of predicting extreme risk has received relatively little attention. The potential benefit of IV is considered within a range of models beginning with the traditional GARCH based approach. Furthermore, a number of novel point process models for forecasting extreme risk are proposed in this paper. Univariate models where IV is included as an exogenous variable are considered along with a novel bivariate approach where extreme movements in IV are treated as another point process. It is found that in the context of forecasting Value-at-Risk, the bivariate models produce the most accurate forecasts across a wide range of scenarios.

Keywords

Implied volatility, Hawkes process, Peaks over threshold, Point process, Extreme events

JEL Classification Numbers

C32, C53, C58.

1 Introduction

Modeling and forecasting extreme losses is a crucially important issue in the management of financial risk. As a result, accurate estimates of risk measures such as Value-at-Risk (VaR) that capture the risk of extreme losses have attracted a great deal of research attention. For a model to be successful in dealing with these extreme loss events it must capture their tendency to cluster in time.

A number of approaches to deal with the clustering of events have been proposed. McNeil and Frey (2000) develop a two stage method where GARCH models are first applied to model the general time variation in volatility with extreme value theory (EVT) techniques then applied to the residuals. Chavez-Demoulin et al. (2005) propose a novel Peaks Over Threshold (POT) approach for modelling extreme events. To deal with event clustering they employ a self-exciting marked point process, specifically a Hawkes process. Under the Hawkes specification, the intensity of the occurrence of extreme events depends on the past events and their associated size or marks. Herrera and Schipp (2013) extend the Hawkes-POT framework of Chavez-Demoulin et al. (2005) in proposing a duration based model to capture the clustering in the extreme loss events.

While they have not been considered in this specific context, option implied volatilities (IV) have been widely used in terms of forecasting volatility. As the volatility of the returns on the underlying asset price is an input into option pricing models, an expectation (risk neutral) of volatility is required before valuing options. While IV is a risk neutral estimate, it is well known that IV indices are negatively correlated with the level of stock market indices and are an important measure of short-term expected risk (see, Bekaert and Wu, 2000; Wagner and Szimayer, 2004; Giot, 2005; Becker et al., 2009; Lin and Chang, 2010; Bekaert and Hoerova, 2014, among others), and have been found to be a useful forecast of physical spot volatility in many studies, see Poon and Granger (2003). Blair et al. (2001) find the inclusion of IV as an exogenous variable in GARCH models to be beneficial in terms of forecasting. While not focusing on forecasting, Becker et al. (2009) show that IV contains useful information about future jump activity in returns, which is likely to reflect extreme movements in prices.

Very few studies have focused on the complex extremal dependence between IV and equity returns. Aboura and Wagner (2014) investigate the asymmetric relationship between daily S&P 500 index returns and VIX index changes revealing the existence of a contemporaneous volatilityreturn tail dependence for negative extreme events though not for positive returns. Peng and Ng (2012) analyse the cross-market dependence between five of the most important equity markets and their corresponding volatility indices, finding evidence of asymmetric tail dependence. Hilal et al. (2011) propose a conditional approach for capturing extremal dependence between daily returns on VIX futures and the S&P500. Their empirical analysis shows that VIX futures returns are very sensitive to stock market downside risk.

In this paper, the analysis moves beyond the role of IV in forecasting total volatility to focus on the link to extreme losses and addresses two main questions.

- 1. How are extreme shocks in an IV index and extreme events in its respective stock market return related?
- 2. Can this relationship be harnessed to provide superior forecasts of extreme returns?

To address these issues, an approach is proposed which utilises IV within intensity based point process models for extreme returns. The first model treats IV as an exogenous variable influencing the intensity and the size distribution of extreme events. A novel alternative view is also proposed based on a bivariate Hawkes model. Extreme movements in IV are treated as events themselves, with their impact on extreme events in equity returns captured through a bivariate Hawkes model. Performance of the proposed methods will be analysed in the context of forecasting extreme losses within a Value-at-Risk framework. The benchmark approach follows both the earlier forecasting literature in that IV is used as an exogenous variable within the GARCH-EVT framework and the bad environments, good environment (BEGE) model of Bekaert and Engstrom (2015).

An empirical analysis is undertaken where forecasts of the risk of extreme returns are generated for five major equity market indices using their associated IV indices. These forecasts are based on GARCH-EVT, BEGE, univariate and bivariate Hawkes models, and take the form of VaR estimates at a range of levels of significance. It is found that GARCH based forecasts which include IV are often inaccurate. Univariate Hawkes and BEGE models where IV is treated as an exogenous variable outperform the GARCH forecasts, though their forecasts do fail a number of tests for VaR adequacy. The bivariate Hawkes models where the timing of past extreme increases in IV are treated as a point process lead to the most accurate forecasts of extreme risk in the widest set of scenarios. The results of this paper show that while IV is certainly of benefit for forecasting extreme risk in equity returns, the framework within which it is used is important. The superior approach is to treat extreme increases in IV as a point process within a bivariate model for extreme returns.

The paper proceeds as follows. Section 2 outlines the traditional GARCH-EVT framework, the BEGE model, and introduces the proposed univariate and bivariate Hawkes point process models. Section 3 describes how VaR forecasts are generated and evaluated. Section 3.1 outlines the equity market and associated IV indices. Section 4 presents in-sample estimation results for the full range of models considered along with the results from tests of forecast accuracy. Section 5 provides concluding comments.

2 Methodology

This section introduces the competing approaches for forecasting extreme losses in the context of VaR predictions. The first is based on the classic GARCH approach where IV is used as an exogenous variable. The specifications considered here are the standard GARCH model of Bollerslev (1986), the GJR-GARCH models of Glosten et al. (1993), and the exponential GARCH (EGARCH) of Nelson (1991). The next approach considered is the BEGE (Bad environment good environment) model of Bekaert and Engstrom (2015) which offers a flexible conditional distribution to describe returns. The approach proposed here utilizes the Hawkes-POT framework introduced in the one-dimensional case by Chavez-Demoulin et al. (2005) which has been employed in a range of empirical applications from modeling equity risk to extreme spikes in electricity prices (Chavez-Demoulin and McGill, 2012; Herrera, 2013; Herrera and Gonzalez, 2014). Here, the one-dimensional approach is extended to include IV as an exogenous variable. A novel bivariate model is also developed to incorporate the intensity of the occurrence of extreme movements in IV. This approach will uncover potential bi-directional linkages between extreme movements in IV and extreme losses. Results from this analysis will reveal whether using IV itself, or the intensity of its extreme movements lead to more precise prediction of the intensity and size of extreme equity market losses.

2.1 Conditional mean and volatility models

The conditional mean of the equity market returns is specified as an Auto Regressive Moving Average (ARMA) process

$$r_t = \mu + \sum_{i=1}^m a_i r_{t-i} + \sum_{j=1}^n b_j \varepsilon_{t-j} + \varepsilon_t.$$
(1)

Where r_t denotes the return on a stock market index at time t, μ a constant, a_i and b_j describe the autoregressive and moving average coefficients, respectively and ε_t denotes the residual term. The residuals are defined by

$$\varepsilon_t = \eta_t \sqrt{h_t}, \qquad \eta_t \sim iid(0,1),$$
(2)

where η_t is the standardized residual and h_t is the conditional variance. The GARCH specifications considered for the conditional variances which include IV as an exogenous variable

$$GARCH(1,1): h_t = \omega + \alpha \varepsilon_{t-1}^2 + \beta h_{t-1} + \gamma I V_{t-1}$$
(3)

$$GJR-GARCH(1,1): h_t = \omega + \alpha \varepsilon_{t-1}^2 + \delta \max(0, -\varepsilon_{t-1})^2 + \beta h_{t-1} + \gamma I V_{t-1}$$
(4)

$$\text{EGARCH}(1,1): \ln h_t = \omega + \alpha \varepsilon_{t-1} + \delta(|\varepsilon_{t-1}| - E|\varepsilon_{t-1}|) + \beta \ln h_{t-1} + \gamma \ln IV_{t-1}.$$
(5)

The GARCH model in (3) corresponds to the standard model of Bollerslev (1986), with $\omega > 0$, $\alpha \ge 0$, $\beta \ge 0$ and $\gamma \ge 0$ so that the conditional variance $h_t > 0$. The model is stationary if $|\alpha + \beta| < 1$ is ensured. The GJR-GARCH specification (4) allows the conditional variance to respond asymmetrically to the sign of past returns by means of the parameter δ . Sufficient conditions for $h_t > 0$ are $\omega > 0$, $\alpha + \delta \ge 0$, $\beta \ge 0$ and $\gamma \ge 0$. Finally, the EGARCH specification (5), allows for asymmetries in volatility if $\alpha \ne 0$ while leverage exists if $\alpha < 0$ and $\alpha < \delta < -\alpha$. To be consistent with the specification of the conditional variance in (5) we also include the IV index in a logarithmic form. These three conditional volatility specifications are estimated assuming a Skew Student-t distribution.¹

The BEGE model of Bekaert and Engstrom (2015) describes the innovations in returns,

$$\varepsilon_t = \sigma_p \omega_{p,t} - \sigma_n \omega_{n,t}, \text{ where}$$

 $\omega_{p,t} \sim \widetilde{\Gamma}(p_t, 1), \text{ and}$
 $\omega_{n,t} \sim \widetilde{\Gamma}(n_t, 1)$ (6)

which is a linear combination of two component shocks, where $\tilde{\Gamma}(k,\theta)$ is a centred gamma distribution with shape and scale parameters, k and θ respectively. The two gamma distributions are assumed to have a constant scale, though time-varying shape parameters, p_t and n_t for the good and bad environments respectively. The shape parameters evolve according to a GJR-GARCH like structure

$$p_{t} = p_{0} + \rho_{p} p_{t-1} + \frac{\sigma_{p}^{+}}{2\sigma_{p}^{2}} \varepsilon_{t}^{2} \mathbf{I}_{\varepsilon_{t-1} \ge 0} + \frac{\sigma_{p}^{-}}{2\sigma_{p}^{2}} (1 - \mathbf{I}_{\varepsilon_{t-1} \ge 0}),$$

$$n_{t} = n_{0} + \rho_{n} n_{t-1} + \frac{\sigma_{N}^{+}}{2\sigma_{n}^{2}} \varepsilon_{t}^{2} \mathbf{I}_{\varepsilon_{t-1} \ge 0} + \frac{\sigma_{n}^{-}}{2\sigma_{n}^{2}} (1 - \mathbf{I}_{\varepsilon_{t-1} \ge 0}).$$
(7)

A version of this model that includes lagged IV as an exogenous variable, with a common coefficient in both positive and negative components (denoted below as BEGE+IV) is also estimated.

¹In a preliminary version of the paper both a conditional Normal, and symmetric student t distribution were also considered. However assuming a skewed student t conditional distribution provides a superior fit to the data. Here the skewness is incorporated into the t-distribution using the method of Fernandez and Steel (1998).

2.2 Conditional intensity models

Marked point processes (MPP) are stochastic processes that couple the temporal clustering arrival times observed in extreme events, with a set of random variables, the so-called marks associated with each event. In EVT, for example, the interest lies in the intensity of extreme event occurrences as well as the distribution of the exceedances over a pre-determined large or extreme threshold. This paper develops two approaches for investigating the role of IV in explaining the intensity and size of extreme loss events. In doing so, the nature of the extreme loss-IV relationship will be revealed.

2.2.1 Univariate Hawkes-POT model

The first point process approach is based on a univariate MPP, specifically the Hawkes-POT model introduced by Chavez-Demoulin et al. (2005) and applied by Chavez-Demoulin and McGill (2012). Here, the Hawkes-POT model is generalised by using the IV index as a co-variate in the conditional intensity process for extreme loss events.

In this context, let $\{(X_t, Y_t)\}_{t\geq 1}$ be a vector of random variables that represent the log-returns of a stock market index and the associated IV derived from options on that index. For ease of subsequent notation, assume returns are multiplied by -1. To determine the conditional intensity of extreme losses, return events whose size exceeds a pre-defined high threshold u > 0 are the focus. This will define a finite subset of observations $\{(t_i, w_i, z_i)\}_{i\geq 1}$, where $t_i \in \mathbb{R}$ corresponds to occurrence times, $w_i \in \mathbb{R}_+$ the magnitude of exceedances (the marks), and $z_i \in \mathbb{R}_+$ a covariate based on the IV index, with $w_i := X_{t_i} - u$, and $z_i := Y_{t_i}$. A general MPP N(t) is proposed satisfying the usual conditions of right-continuity $N(t) := N(0, t] = \sum_{i\geq 1} \mathbb{1}\{t_i \leq t, w_i = w, z_i = z\}$ with past history or natural filtration $\mathcal{H}_t = \{(t_i, w_i, z_i) \ \forall i : t_i < t\}$ that includes times, marks and the covariates. According to the standard definition of a MPP, it may be characterized by means of its conditional intensity function

$$\lambda\left(t, w \mid \mathcal{H}_{t}\right) = \lambda_{q}\left(t \mid \mathcal{H}_{t}\right) g\left(w \mid \mathcal{H}_{t}, t\right), \tag{8}$$

which, broadly speaking describes the probability of observing a new event in the next instant of time conditional on the history of the process.

There are two components to the intensity of the MPP, a ground process $N^g(t) := \sum_{i\geq 1} \mathbb{1}\{t_i \leq t\}$ with conditional intensity $\lambda_g(t \mid \mathcal{H}_t)$ which characterizes the rate of the extreme events over time, and the process for the marks, whose density function $g(w \mid \mathcal{H}_t, t)$ is conditional on the history of the process and time t. Observe that the covariate z_i does not directly enter into the definition of the conditional intensity in equation (8) even though it appears to be another mark in addition to w_i contained in the available information set, \mathcal{H}_t . Instead, the covariate z_i provides extra information to explain the behaviour of the process without being directly involved in the determination of likelihood in this stochastic process.

The conditional intensity $\lambda_g(t \mid \mathcal{H}_t)$ is characterized by the branching structure of a Hawkes process with an exponential decay function

$$\lambda_g \left(t \mid \mathcal{H}_t \right) = \nu + \vartheta \sum_{i: t_i < t} e^{\psi w_i + \rho z_i} \phi e^{-\phi(t - t_i)}, \tag{9}$$

where $\nu \geq 0$ is the background intensity that accounts for the intensity of exogenous events independent of the internal history \mathcal{H}_t , the branching coefficient $\vartheta \geq 0$ describes the frequency with which new extreme events arrive, the parameter $\psi \in \mathbb{R}$ and $\rho \in \mathbb{R}$ determine the contribution of the mark w_i and covariate z_i to the conditional intensity of the ground process, and $\phi > 0$ is a decay parameter. The exponential functions inside the sum define the impact $f(w, z) = e^{\psi w + \rho z}$, and kernel decay function $h(t - t_i) = \phi e^{-\phi(t-t_i)}$ that controls how offspring are generated by first order extreme events which represents the main source of clustering in the model. This process is described as self-exciting as the occurrence times and marks of past extreme events may make the occurrence of future extreme events more probable through the dependance on the history, \mathcal{H}_t .

To estimate risk measures such as VaR, an assumption regarding the probability distribution function of the most extreme return events, w_i conditional on the event that X_{t_i} exceeds the threshold u > 0 must be made. Motivated by the Pickands–Balkema–de Haan's theorem, ² the extreme losses are assumed to follow a conditional Generalized Pareto Distribution (GPD) with density function given by

$$g(w \mid \mathcal{H}_t, t) = \begin{cases} \frac{1}{\kappa(w \mid \mathcal{H}_t, t)} \left(1 + \xi \frac{w}{\kappa(w \mid \mathcal{H}_t, t)} \right)^{-1/\xi - 1} &, \quad \xi \neq 0\\ \frac{1}{\kappa(w \mid \mathcal{H}_t, t)} \exp\left(-\frac{w}{\kappa(w \mid \mathcal{H}_t, t)} \right) &, \quad \xi = 0, \end{cases}$$
(10)

where ξ is the shape parameter and $\kappa (w | \mathcal{H}_t, t)$ is a scale parameter specified as a self-exciting function of the arrival times of new extreme events and their sizes

$$\kappa \left(w \mid \mathcal{H}_t, t \right) = \kappa_0 + \kappa_1 \sum_{i: t_i < t} e^{\psi w_i + \rho z_i} \phi e^{-\phi(t - t_i)}.$$

Under this specification, $\kappa_0 \geq 0$ represents the baseline level for the scale, while $\kappa_1 \geq 0$ is an impact parameter related to the influence of new extreme event arrivals. The shape parameter is assumed to be constant through time due to the sparsity of events in the tail of the distribution which makes estimation of time-varying scale challenging (as evident in Chavez-Demoulin et al.,

²See Pickands (1975) and Balkema and $\overline{\text{De}}$ Haan (1974).

2005; Santos and Alves, 2012; Herrera, 2013).

The log-likelihood for the univariate Hawkes-POT model given a set of events $\{(t_i, w_i, z_i)\}_{i=1}^{N(t)}$ observed in the space $(0, T] \times [u, \infty)$ is obtained combining the conditional intensity (8) and the density of the marks (10) as follows

$$\ell = \sum_{i=1}^{N(T)} \ln \lambda_g (t_i \mid \mathcal{H}_{t_i}) - \int_0^T \lambda_g (s \mid \mathcal{H}_s) \, ds + \sum_{i=1}^{N(T)} \ln g (w_i \mid \mathcal{H}_{t_i}, t_i)$$
(11)
$$= \sum_{i=1}^{N(T)} \ln \left(\nu + \vartheta \sum_{i:t_i < T} e^{\psi w_i + \rho z_i} \phi e^{-\phi(T-t_i)} \right) - \left\{ \nu T + \vartheta \sum_{i:t_i < T} e^{\psi w_i + \rho z_i} \left(1 - e^{-\phi(T-t_i)} \right) \right\}$$
$$- \left[(1/\xi + 1) \sum_{i=1}^{N(T)} \left\{ \ln \kappa (w_i \mid \mathcal{H}_{t_i}, t_i) + \ln (1 + \xi w_i / \kappa (w_i \mid \mathcal{H}_{t_i}, t_i)) \right\} \right]$$

assuming for ease of the exposition that $\xi \neq 0$. The resulting estimates are consistent, asymptotically normal and efficient, with standard errors obtained via the Fisher information matrix (Ogata, 1978).

2.2.2 Bivariate Hawkes-POT model

The novel bivariate approach proposed here moves beyond simply including IV as an exogenous covariate. Extreme increases in IV are treated as a second MPP and represent the second dimension in a bivariate model in addition to the extreme stock market losses. In this bivariate model, the marks can influence the evolution of its respective ground process and vice versa, offering a framework to examine the impact of IV events on extreme stock market losses.

The bivariate MPP is defined as a vector of point processes $\mathbf{N}(t) : \{N_1(t), N_2(t)\}$, where the first point process $N_1(t)$ is defined through the pairs $\{(t_i^1, w_i)\}_{i \ge 1}$; the subset of extreme events in the negative log-returns of the stock market occurring at time t_i^1 over a high threshold $u_1 > 0$, with $w_i := X_{t_i^1} - u_1$. Similarly, the second point process $N_2(t)$ is defined by the pairs of events $\{(t_i^2, z_i)\}_{i\ge 1}$ with $z_i := Y_{t_i^2} - u_2$, which also characterizes the subset of extreme events occurring in IV at time t_i^2 over a high threshold $u_2 > 0$. $\mathcal{H}_t = \{(t_i^1, w_i), (t_j^2, z_j) \; \forall i, j : t_i^1 < t \land t_j^2 < t\}$ denotes the combined history over all times and marks. This bivariate MPP includes a bivariate ground process $N_k^g(t) := \sum_{i\ge 1} \mathbbm{1}\{t_i^k \le t\}$ with conditional intensities

$$\lambda_{g}^{1}(t \mid \mathcal{H}_{t}) = \nu_{1} + \vartheta_{11} \sum_{i:t_{i}^{1} < t} e^{\psi_{1}w_{i}} \phi_{1} e^{-\phi_{1}\left(t-t_{i}^{1}\right)} + \vartheta_{12} \sum_{i:t_{i}^{2} < t} e^{\rho_{1}z_{i}} \phi_{2} e^{-\phi_{2}\left(t-t_{i}^{2}\right)}$$
(12)
$$\lambda_{g}^{2}(t \mid \mathcal{H}_{t}) = \nu_{2} + \vartheta_{21} \sum_{i:t_{i}^{1} < t} e^{\psi_{2}w_{i}} \phi_{1} e^{-\phi_{1}\left(t-t_{i}^{1}\right)} + \vartheta_{22} \sum_{i:t_{i}^{2} < t} e^{\rho_{2}z_{i}} \phi_{2} e^{-\phi_{2}\left(t-t_{i}^{2}\right)}$$

where $\nu_k \geq 0$ are the background intensities, the branching coefficients $\vartheta_{jk} \geq 0$ describe the influence that dimension k will have on dimension j, the parameters $\psi_k \geq 0$ and $\rho_k \geq 0$ determine the contribution of the size of the extremes occurring at the returns and IV to the conditional intensity of the ground process, and $\phi_k > 0$ are again the decay parameters. Thus, the impact functions $f_k(w) = e^{\psi_k w}$ and $f_k(z) = e^{\rho_k z}$, and the exponential decay kernel function $h_k (t - t_i^k) = \phi_k e^{-\phi_k (t - t_i^k)}$ account for mutual and cross excitation.

A key feature of the proposed bivariate MPP is that it only includes a true mark for the point process of the stock market returns, with the distribution of the marks for the IV events always set to unity, $g(z | \mathcal{H}_t, t) = 1$ implying the conditional intensity for these events is

$$\lambda^2 \left(t, z \mid \mathcal{H}_t \right) = \lambda_q^2 \left(t \mid \mathcal{H}_t \right). \tag{13}$$

This assumption is maintained as the focus is on estimating measures of risk for the stock market returns given the behavior of IV at extreme levels (i.e., conditional intensity, occurrence times and size of extreme events in IV). To achieve this it is not necessary to model the distribution of the extreme IV events thus reducing possible estimation error.

Similar to the univariate MPP, a generalized Pareto density for the stock market returns as in (10), is used again but with conditional scale parameter

$$\kappa \left(w \mid \mathcal{H}_{t}, t \right) = \kappa_{0} + \kappa_{1} \sum_{i:t_{i}^{1} < t} e^{\psi_{1}w_{i}} \phi_{1} e^{-\phi_{1}\left(t - t_{i}^{1}\right)} + \kappa_{12} \sum_{i:t_{i}^{2} < t} e^{\rho_{1}z_{i}} \phi_{2} e^{-\phi_{2}\left(t - t_{i}^{2}\right)}.$$
(14)

Under this specification $\kappa_{12} \ge 0$ is an impact parameter related to the influence of the arrival times and size of extreme events occurring in the IV index.

Given the occurrence of pairs of observations $\{(t_i^1, w_i)\}_{i=1}^{N_1(T)}$ and $\{(t_i^2, z_i)\}_{i=1}^{N_2(t)}$ in a set $(0, T] \times [u_1, \infty)$ and $(0, T] \times [u_2, \infty)$ respectively, the log-likelihood for this bivariate point process is obtained linking the bivariate conditional intensity for the ground process (12) and the density for the marks of the stock market returns (10) with scale parameter defined by (14).

$$\ell = \sum_{k=1}^{2} \left\{ \sum_{i=1}^{N^{k}(T)} \ln \lambda_{g}^{k} \left(t_{i}^{k} \mid \mathcal{H}_{t_{i}^{k}} \right) - \int_{0}^{T} \lambda_{g}^{k} \left(s \mid \mathcal{H}_{s} \right) ds \right\} + \sum_{i=1}^{N^{1}(T)} \ln g \left(\mathcal{H}_{t_{i}^{1}}, t_{i}^{1} \right)$$

$$= \sum_{k=1}^{2} \sum_{i=1}^{N^{k}(T)} \ln \left(\nu_{k} + \vartheta_{k1} \sum_{j:t_{j}^{1} < t_{i}^{k}} e^{\psi_{k} w_{j}} \phi_{1} e^{-\phi_{1} \left(t_{i}^{k} - t_{j}^{1} \right)} + \vartheta_{k2} \sum_{j:t_{j}^{2} < t_{i}^{k}} e^{\rho_{k} z_{j}} \phi_{2} e^{-\phi_{2} \left(t_{i}^{k} - t_{j}^{2} \right)} \right)$$

$$- \left\{ (\nu_{1} + \nu_{2})T + \sum_{k=1}^{2} \left\{ \vartheta_{k1} \sum_{j:t_{j}^{1} < T} e^{\psi_{k} w_{j}} \left(1 - e^{-\phi_{k} \left(T - t_{j}^{1} \right)} \right) + \vartheta_{k2} \sum_{j:t_{j}^{2} < T} e^{\rho_{k} z_{j}} \left(1 - e^{-\phi_{2} \left(T - t_{j}^{2} \right)} \right) \right) \right\} \right\}$$

$$- \left[(1/\xi + 1) \sum_{i=1}^{N^{1}(T)} \left\{ \ln \kappa \left(w_{i} \mid \mathcal{H}_{t_{i}^{1}}, t_{i}^{1} \right) + \ln \left(1 + \xi w_{i} / \kappa \left(\mathcal{H}_{t_{i}^{1}}, t_{i}^{1} \right) \right) \right\} \right],$$

$$(15)$$

assuming once again for ease of the exposition that $\xi \neq 0$.

Following Embrechts et al. (2011), in the following preposition we state some weak conditions to determine the existence of a Hawkes-POT process with stationary increments and asymptotically stationary conditional ground intensity.

Preposition 1. (Stationarity) The conditional ground intensities defined in (9) and (12) are asymptotically stationary under the following stability conditions

• Univariate model: Define $h(s) := \phi e^{-\phi(s)}$ and $f(w, z) := e^{\psi w + \rho z}$ as the decay kernel and impact function, respectively. Then, given that the decay kernel function satisfies $\int_0^\infty h(s) ds = 1$, and the expectation of impact function exists $\mathbb{E}[f(w, z)] = \mu_{wz}$, the univariate model defined in (9) is asymptotically stationary, if and only if,

$$0 < \vartheta \mu_{wz} < 1$$

Bivariate model: Define h_k (s) := φ_ke^{-φ_k(s)} as the decay function satisfying ∫₀[∞] h_k (s) ds = 1 for k = 1, 2, and f_k(w) := e^{ψ_kw} and f_k(z) := e^{ρ_kz} as the impact functions of the marks and covariates with expectations given by E [f_k(w)] = μ^k_w and E [f_k(z)] = μ^k_z, respectively. In addition, denoting M := {(μ^k_w, μ^k_z) : k ∈ {1,2}} and Q := {ϑ_{jk} : j, k ∈ {1,2}} the (2 × 2) as the matrix representations of the expectations and branching coefficients. The bivariate model defined in (12) is asymptically stationary, if and only if, the spectral radius of the matrix M ∘ Q is less than one, i.e.,

$$Spr(M \circ Q) := \max\left\{ |\varphi| : det(M \circ Q - \varphi \mathbf{1}_2) = 0 \right\} < 1,$$

where $\mathbf{1}_2$ is the (2×2) identity matrix, φ are the eigen values of $M \circ Q$, and \circ denotes the Hadamard product.

3 Generating and evaluating forecasts conditional risk measures

The accuracy of the forecasts of extreme events will be analysed in the context of conditional risk measures. How these risk measures are generated from the various approaches will now be described. VaR^t_{α} is the VaR computed at day t-1 for the negative log-return X_t as follows

$$1 - \alpha = P\left(X_t > VaR_{\alpha}^t \mid \mathcal{H}_t\right)$$

where the equality above assumes a continuous distribution for X_t . Most financial return series exhibit stochastic volatility, autocorrelation, and fat-tailed distributions limiting the direct estimation of the VaR. For this reason, under the traditional benchmark approach the first stage consists of filtering the returns series with a ARMA-GARCH process such that the residuals are closer to iid. Given the assumed dynamics for the conditional mean of returns in (1), and the conditional volatility proposed in (2) the following model for the returns is obtained

$$X_t = \mu + \sum_{i=1}^m a_i X_{t-i} + \sum_{j=1}^n b_j \varepsilon_{t-j} + \varepsilon_t, \qquad (16)$$

where $\varepsilon_t = \eta_t \sqrt{h_t}$ and h_t is the stochastic conditional variance, $h_t \in \mathcal{H}_t$. The autoregressive specifications for the conditional variances including the GARCH, GJR-GARCH and EGARCH are shown in (3), (4) and (5), respectively. In the second stage, the corresponding VaR at the α confidence level of the assumed distribution of the residuals η_t , i.e., $VaR_\alpha(\eta_t)$: inf $\{x \in \mathbb{R} : P(\eta_t > x) \leq 1 - \alpha\}$ is used to obtain estimates for the conditional VaR for the returns. Observe that η_t are iid, and therefore $VaR_\alpha(\eta_t) = VaR_\alpha(\eta_{t-1}) = \cdots = VaR_\alpha(\eta_{t-j}) =:$ $VaR_\alpha(\eta)$, implying that (16) can be rewritten as follows

$$VaR^t_{\alpha} = \mu_{t-1} + VaR_{\alpha}(\eta)\sigma_{t-1},$$

where $\mu_{t-1} = \mu + \sum_{i=1}^{m} a_i X_{t-i} + \sum_{j=1}^{n} b_j \varepsilon_{t-j}$ and $\sigma_{t-1} = \sqrt{h_t}$. Note that the history \mathcal{H}_t in this type of model is generated in a discrete time framework, contrary to the filtration generated by the point process approach where time is continuous. Therefore all information relating to the stochastic process prior (but not at) to time t can be included.

VaR forecasts from the BEGE models are obtained by numerically inverting the BEGE cumulative distribution (used to numerically evaluate the probability distribution function and hence the likelihood) function at the required α confidence level, given forecasts of p_t and n_t . By doing so, this takes into account not only the conditional variance, but also the higher moments of the distribution when generating the VaR forecast.

The two (univariate and multivariate) Hawkes-POT models described in Section 2.2 can also be directly used to estimate VaR. The advantage of this approach is that it avoids the filtering of returns and the use of EVT. Observe that the conditional probability that the next daily return X_t will exceed the threshold u > 0 given that X_{t-1} has already exceeded this threshold is given by

$$\mathbb{P}(X_t > u \mid \mathcal{H}_t) = 1 - \mathbb{P}\{N([t-1,t) = 0 \mid \mathcal{H}_t)\}\$$

= $1 - \exp\left(-\int_{t-1}^t \lambda_g(s \mid \mathcal{H}_s) ds\right),\$
 $\approx \lambda_g(t \mid \mathcal{H}_t).$ (17)

On the other hand, the conditional probability of this event, given that the high threshold u > 0has been exceeded, exceeds an even higher threshold (u + x) > 0 is modeled using a generalized Pareto distribution

$$\mathbb{P}\left(X_t - u > x \mid X_t > u, \mathcal{H}_t\right) = \overline{G}\left(x - u \mid \mathcal{H}_t, t\right), \tag{18}$$

where $\overline{G}(x - u \mid \mathcal{H}_t, t)$ corresponds to the survival function of the cumulative distribution function of (10). One can demonstrate that for Hawkes-POT models, the probability that the next daily return X_t will exceed the VaR at the α confidence level is a solution to the equation $\mathbb{P}(X_t > \operatorname{VaR}^t_{\alpha} \mid \mathcal{H}_t) = 1 - \alpha$, or alternatively,

$$\mathbb{P}\left(X_t > \operatorname{VaR}_{\alpha}^t \mid \mathcal{H}_t\right) = \mathbb{P}\left(X_t > u \mid \mathcal{H}_t\right) \mathbb{P}\left(X_t - u > \operatorname{VaR}_{\alpha}^t - u \mid X_t > u, \mathcal{H}_t\right).$$
(19)

Thus, given the conditional intensity for the ground process (17) and the distribution for the marks (18), a solution to (19) leads to a prediction of the VaR in the next instant at the α confidence level

$$\operatorname{VaR}_{\alpha}^{t} = u + \frac{\kappa \left(w \mid \mathcal{H}_{t}\right)}{\xi} \left\{ \left(\frac{\lambda_{g}(t \mid \mathcal{H}_{t})}{1 - \alpha}\right)^{\xi} - 1 \right\}.$$
(20)

Depending on the approach, univariate or bivariate, the ground conditional intensity in (20) is replaced with either (9) or (12). The same occurs for the scale parameter.

To assess the accuracy of the competing approaches for predicting VaR at different confidence levels, the following set of statistical tests, based on both long-standing and quite new approaches are employed. For further details relating to these test see Christoffersen (1998); Engle and Manganelli (2004); Ziggel et al. (2014). Let $\{I_t(\alpha)\}_{t=1}^n$ be a vector of ex-post indicator variables of VaR exceptions taking the value 1 if $X_t > VaR_{\alpha}^t$ and 0 if $X_t \leq VaR_{\alpha}^t$ at time t at the VaR coverage probability α . In addition, define the variable $Hit_t(\alpha) = I_t(\alpha) - \alpha$ as the de-meaned hits of exceptions.

The first test is the unconditional coverage test (LR_{uc}) introduced by Kupiec (1995). In short, this test is concerned with whether or not the reported VaR exceptions occur more (or less) frequently than $\alpha \times 100\%$ of the time. The second test examines the independence of these exceptions (LR_{ind}) using a Markov test. The third test is the conditional coverage test (LR_{cc}) , which is a combination of the previous two tests. The key point of this test is that an accurate VaR measure must exhibit both the independence and unconditional coverage properties. The next two tests are the regression based Dynamic Quantile tests introduced by Engle and Manganelli (2004), where the regressors are the lagged Hit_t in the Dynamic Quantile Hit (DQ_{hit}) test, whereas the Dynamic Quantile VaR (DQ_{VaR}) also includes past VaR estimates as an explanatory variable.

Recently, Ziggel et al. (2014) proposed a new set of tests that, beside testing the unconditional coverage and independence of VaR exceptions, importantly they also test that exceptions are identically distributed. Another advantage of these new tests is that all critical values for these tests are distribution free and can be obtained utilizing Monte Carlo simulations, allowing one- and two-tailed tests. Under this framework, the null hypothesis of unconditional coverage test is satisfied if the expectation of VaR exceptions is equal on average to α , i.e., $H_0: \mathbb{E}\left[\frac{1}{n}\sum_{t=1}^{n} I_t(\alpha)\right] = \alpha$. They propose the statistic:

$$MC_{uc} = \sum_{t=1}^{n} I_t(\alpha) + \epsilon, \qquad (21)$$

where ϵ is a continuous random variable with a small variance designed to help to break ties between test values.

To test for iid VaR exceptions, Ziggel et al. (2014) utilizes the fact that waiting times between VaR exceptions should be geometrically distributed. In particular, they propose to test the null hypothesis H_0 : $\mathbb{E}[t_i - t_{i-1}] = \frac{1}{\alpha}$, by looking at the squared waiting times between VaR exceptions, which are better suited to detect exceptions which occur in clusters:

$$MC_{iid,m} = t_1^2 + (n - t_m)^2 + \sum_{i=2}^m (t_i - t_{i-1})^2 + \epsilon,$$
(22)

where m is the sum of observed VaR exceptions and t_1, \ldots, t_m describe the occurrence times of VaR exceptions. Note, that the value of this statistic increases as the waiting times exhibit a greater degree of correlation among them. Thus, this test is very useful for detecting clustering among the VaR exceptions implying a poor prediction. The final test corresponds to a conditional coverage tests whose specification is given by:

$$MC_{cc,m} = a \cdot f(MC_{uc}) + (1-a) \cdot g(MC_{iid,m}), \ 0 \le a \le 1,$$
(23)

where $f(MC_{uc}) = \left|\frac{MC_{uc}/n-p}{p}\right|$ and $g(MC_{iid,m}) = \frac{MC_{iid,m}-\hat{r}}{\hat{r}} \mathbb{1}_{\{MC_{iid,m} \geq \hat{r}\}}$ measure the difference between the expected and observed proportions of VaR exceptions, and sum of squared waiting times, respectively. The parameter a is a weighting factor that can be chosen according to an individuals preference toward the importance of either the iid property or the correct unconditional coverage of the exceptions. In the subsequent empirical analysis, the importance of both properties are treated equal, and only results for a = 0.5 are presented. In the last term, \hat{r} denotes an estimator of the expected value of the statistic $MC_{iid,m}$ under the null hypothesis. All critical values of these tests statistic are obtained utilizing 10,000 Monte Carlo simulations of the finite sample null distribution.

To ensure that the test statistics follow a continuous distribution, a continuous random variable with an arbitrarily small variance, $\epsilon \sim N(0, 1e^{-6})$ is used in all applications³. For further details on the last three tests see Ziggel et al. (2014).

Backtesting considers whether each individual model produces VaR forecasts that are adequate in their own right and satisfy the coverage, independence properties. While this is important, these tests do not allow conclusions to be drawn on which model produces the most accurate VaR forecast. First, to directly measure forecast accuracy, the asymmetric quantile loss function

$$\ell\left(X_t, VaR_{\alpha}^t\right) = \left(I_t\left(\alpha\right) - \alpha\right)\left(X_t - VaR_{\alpha}^t\right) \tag{24}$$

proposed by González-Rivera et al. (2004) is used. $I_t(\alpha)$ is again the indicator function taking the value 1 when an exception occurs at α significance and 0 otherwise. The motivation behind this loss function is very intuitive in the context of risk management since VaR exceptions are penalised more heavily. Such a loss function would underly a fairly broad class of economic applications involving capital allocation in response to risk forecasts.

Given the quantile loss function in equation 24, significant differences in VaR forecast performance will be assessed using the Model Confidence Set (MCS) introduced by Hansen et al. (2011). The MCS approach avoids the specification of a benchmark model, and starts with a full set of candidate models $\mathcal{M}_0 = \{1, ..., m_0\}$. All loss differentials, $d_{ij,t}$, using equation 24, between models *i* and *j* are computed and the null hypothesis, H_0 : $E(d_{ij,t}) = 0$ is tested for each pair. If H_0 is rejected at the significance level α_M , the worst performing model is removed and the process continues until non-rejection occurs with the set of surviving models being the

³According to Ziggel et al. (2014) the finite sample accuracy of the test statistics are not greatly affected by the choice of a continuous probability distribution function for ϵ , provided that its variance is small.

MCS, $\widehat{\mathcal{M}}^*_{\alpha_M}$. If a fixed significance level α_M is used at each step, $\widehat{\mathcal{M}}^*_{\alpha_M}$ contains the best model from \mathcal{M}_0 with $(1 - \alpha_M)$ confidence. The null hypothesis is tested by means of the range statistic for combining individual *t*-statistics from the pairwise comparison of forecasts. An estimate of the asymptotic variance of the pairwise loss differentials is obtained from a bootstrap procedure described in Hansen et al. (2003). Reported p-values are corrected to ensure consistency through the iterative testing framework. See Hansen et al. (2003) and Hansen et al. (2011) for more detail. In all subsequent empirical results, a level of 95% confidence will be used in the MCS analysis.

The estimation of VaR for horizons longer than one day is an important issue in determining financial risk. However, the extension of the Hawkes-POT model from a single prediction period to a longer horizon is not a trivial exercise. This is due to the dynamic specification of how extreme events occur over time, based on point process theory where the intensity is characterized by means of a stochastic counting process. As a final measure of the performance of the bivariate point process models, a simple attempt is made to obtain multi-period VaR estimates and examine whether they satisfy the standard tests discussed earlier. This is achieved by scaling the one period VaR by a factor $h^{\hat{\xi}_u}$

$$VaR^t_{\alpha} \approx h^{\hat{\xi_u}} VaR^{t+h}_{\alpha}$$

where h is the horizon time and $\hat{\xi}_u$ is the unconditional shape parameter obtained from the raw log-returns. The approach used here is based on EVT suggesting that the estimation of long-term VaR is actually possible for fat tailed distributions (see Danielsson and De Vries, 2000; Cotter, 2007, for empirical applications of this approach). The major advantage of this simple approach, is that besides the estimation of the unconditional shape parameter $\hat{\xi}_u$, there is not need to re-estimate any additional parameters. Although VaR accuracy for multi-periods is complicated by the fact that the VaR exceptions are intrinsically autocorrelated, the unconditional coverage test (LR_{uc}) for a horizon period of 5 and 10 days accounting autocorrelation in the test. Given the overlapping nature of the multi-period forecasts none of the more complex tests are undertaken.

3.1 Data

The data consists of daily returns for the S&P 500, Nasdaq, DAX 30, Dow Jones and Nikkei stock market indices, and their respective IV indices, VIX, VXN, VDAX, VXD, and VXJ. As the focus is on extreme increases in IV, events will be defined on daily log-changes in IV, $\Delta IV_t = \ln (IV_t/IV_{t-1})$ for each market. All data series used here are obtained from Bloomberg. For each pair of stock market index and IV, the longest sample of data available is collected (S&P

	S&P 500	VIX	DAX 30	VDAX	Dow Jones	VXD	Nikkei	VXJ	Nasdaq	VXN
mean	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
sd	0.01	0.03	0.01	0.05	0.01	0.06	0.02	0.06	0.02	0.05
min	-0.09	-0.15	-0.09	-0.27	-0.08	-0.33	-0.12	-0.42	-0.10	-0.31
max	0.11	0.22	0.11	0.31	0.11	0.53	0.13	0.58	0.11	0.36
skewness	-0.24	0.65	-0.10	0.68	-0.08	0.64	-0.32	1.45	-0.07	0.57
kurtosis	11.65	7.31	7.40	6.81	9.97	7.05	9.11	16.08	7.41	6.74
Ljung-Box	42.26*	109.82^{*}	24.51*	37.77^{*}	39.74^{*}	59.09^{*}	9.43^{***}	31.01*	18.71*	31.41*
Jarque Bera	18916.57^{*}	5118.41*	4292.14^{*}	3637.99^{*}	7639.51^{*}	2837.02^{*}	5806.37^{*}	27554.81^*	2632.06	2070.05^{*}
ADF test	-18.20*	-19.81*	-16.94*	-17.83^{*}	-15.31^{*}	-16.49*	-15.38*	-15.97*	-14.54	-16.80^{*}
Start of	02/01	1/99	02/0	1/92	02/01	/98	05/0	01/98	01/0	01/00
sample period										
Extreme	60	6	53	33	37	9	3	70	3	26
events										
Comovements	34	.0	30)8	20	1	1	97	1	.60

Table 1: Descriptive statistics for the daily stock market returns and IV log-changes. The Ljung-Box statistics are significant for a lag of 5 trading days. *, **, *** represent significance at 1%, 5% and 10% levels, respectively. All the samples end in December 31, 2013.

500: 02/01/90, DAX: 02/01/92, Dow Jones: 02/1/98, Nikkei: 05/01/98, Nasdaq: 01/01/00), with all series ending December 31, 2013. The period ending 30 December 2011 is used for estimation while 2012-2013 is used for backtesting.

The VIX index was the first widely published IV index upon which trading was developed. IV for other U.S. indices (VXN and VXD) and both the European (VDAX) and Japanese markets (VXJ) all follow the same principle as the VIX. The VIX index was developed by the Chicago Board of Options Exchange from S&P 500 index options to be a general measure of the market's estimate of average S&P 500 volatility over the subsequent 22 trading days. It is derived from out-of-the-money put and call options that have maturities close to the fixed target of 22 trading days. For technical details relating to the construction of the VIX index, see CBOE (2003).

Descriptive statistics for each series is given in Table 1. It is clear that for all markets, the sample standard deviation of changes in IV are much larger than the corresponding equity index returns. All series exhibit high levels of kurtosis, stock market returns are negatively skewed and changes in IV are positively skewed. Indeed, none of the series analysed is normally distributed based on the Jarque-Bera statistic. The Ljung-Box statistics reject the null of no autocorrelation at a lag of 5 trading days for all series. Augmented Dickey–Fuller tests for the presence of unit roots show that all time series are stationary at 1% significance level.

Extreme movements in stock market returns or changes in IV are events for which the probability is small. Here, an extreme event is defined as one that belongs to the 10% of the most negative returns in stock markets, or to the 10% of the most positive log-changes in IV. Table 1 reports the number of extreme events occurring in stock market returns and IV indices, independently and simultaneously. Of these, between 49% and 57% of the extreme events are joint events where extreme movements in equity returns and IV occur simultaneously, reflected in the comovements reported in the bottom row. Figure 1 gives an overview of the comovement at extreme levels in stock market returns and IV. The plots in the left column show the market returns and IV indices, with bars under the IV indicating the occurrence of the most negative extreme returns given the occurrence of a positive extreme change in IV. In a similar fashion to volatility itself, these extreme events tend to cluster through time. The centre column of plots shows the relationship between changes in IV and stock market returns. Overall, there is a clear negative relationship between the two series. reflecting the commonly observed asymmetry in the equity return-volatility relationship. The occurrence of extreme events, negative market returns and positive changes in IV, which are of central interest here are represented by the black dots.

An alternative approach for measuring extremal comovement is the extremogram introduced by Davis et al. (2009). This is a flexible conditional measure of extremal serial dependence, which makes it particularly well suited for financial applications. The plots in the right column in Figure 1 show the sample extremograms for the 10% of the most negative stock returns conditional on the 10% of the most positive log-changes on the IV indices at different lags. The interpretation of the extremogram is similar to the correlogram, given that the IV index has experienced an extreme positive change at time t, the probability of obtaining a negative extreme shock in stock market returns at time t + k is reflected by the solid vertical lines in the sample extremogram for each lag k. The grey lines represent the .975 (upper) and .025 (lower) confidence interval estimated using a stationary bootstrap procedure proposed by Davis et al. (2012), while the dashed line corresponds to this conditional probability under the assumption that extreme events in both markets are occurring independently (for more details on the estimation refer to Davis et al., 2012).⁴ Observe that the speed of decay of the sample extremograms for all five markets is extremely slow, revealing that the dependence of extreme movements in returns on IV shocks is significant out to about 10 lags in most cases.

4 Empirical results

This section presents both in-sample estimation results in Section 4.1, and comparisons of forecast performance in terms of risk prediction in Section 4.2.

4.1 Estimation results

Estimation results discussed in this section are based on data up to 30 December 2011. To begin, Table 2 reports the estimation results for the various GARCH and BEGE specifications. Results for models using a skew t-distribution are reported, assuming either a conditional normal, or

 $^{^{4}2000}$ pseudo-series are generated for the estimation of the extremograms utilizing a stationary bootstrap with resampling based on block sizes from a geometric distribution with a mean of 200.



Figure 1: (left column) Extreme negative returns (grey color) and IV log-changes (blue color) to display the asymmetric association between them. (middle column) Scatter plot of IV log-changes and stock market returns. The 10% of the most extreme negative (positive) stock market returns (IV log-changes) are displayed in grey color. (right column) Sample extremograms for the 10% of the most negative stock returns conditional to the 10% of the most positive log-changes on the IV indices at different lags. Grey lines represent the .975 (upper) and .025 (lower) confidence interval estimated using a stationary bootstrap procedure proposed by Davis et al. (2012), while the dashed line (blue color) corresponds to this conditional probability under the assumption that extreme events between both markets occur independently.

symmetric student t- distribution leads to inferior results and hence the results are not reported here. Estimates of the GARCH coefficients reveals a number of common patterns. For models that do not include IV as an exogenous regressor, estimates of the β coefficient are in excess of 0.9 indicating a strong degree of volatility persistence. When IV is included, γ is found to be significant and the presence of IV helps explain a degree of the persistence in many of the cases with the estimate of β falling. As is to be expected, estimates of the asymmetry coefficient δ in both the GJR-GARCH and EGARCH models are significant. Conditionally, returns are found to exhibit relatively heavy tails with estimates of the ν falling between 7 and 15. In all cases, estimates of the skew parameter ψ significantly less than one indicate that returns are conditionally negatively skewed, supporting the choice of the skewed-t distribution. Of the competing models, EGARCH including IV offer the best model fit for all markets. Overall, the fit of the BEGE models are close to both the GJR-GARCH and EGARCH models. While both positive and negative components of volatility are found to be persistent, the negative component exhibits less persistence than the corresponding positive component ($\rho_n < \rho_p$), a result consistent with the findings of Bekaert and Engstrom (2015). In three of the markets, the impact of IV on the shape parameters, p_t and n_t are significant and positive with the final two still positive though not significant.

Three versions of the univariate model in (9) are estimated. Model 1 is the full model with marks $(\psi > 0)$ and IV $(\rho > 0)$. Model 2 only includes marks $(\psi > 0)$ restricting $\rho = 0$. Model 3 includes neither marks nor covariates and restricts $\psi = 0$ and $\rho = 0$. Table 3 reports the estimation results for the three univariate models. In all cases, the unrestricted Model 1 offers the best overall fit. Estimates for ψ are significant in all instances, reflecting the importance of the size of past marks for future intensity. On the other hand, estimates of ρ are strongly significant only in the S&P500 and Nikkei markets meaning that the level of IV is only important for explaining the intensity of extreme events in these two markets. While ρ is marginally significant for the DAX, it is insignificant for the other two remaining markets.

Similar to the univariate case, four versions of the bivariate model are estimated. The ground intensities under Model 1 are generated by the full unrestricted model in equation (12) and contain the past times and marks of both extreme return and IV events, with ψ_1 , ψ_2 , $\rho_1 \rho_2 > 0$, and with the scale of the return marks specified in equation (14). Model 2 also includes the past times and marks of both extreme return and IV events, with the restriction that $\psi_1 = \psi_2$ and $\rho_1 = \rho_2$ with the scale only driven by the arrival times and size of the past return events ($\kappa_{12} = 0$). Model 3 contains the times and marks of return events ($\psi_1, \psi_2 > 0$) but only the times of past IV events (i.e., $\rho_1 = \rho_2 = 0$) with the scale only driven by the size of the past return events. The ground intensities under Model 4, are restricted to contain the times of past return and IV events, $\psi_1 = \psi_2 = 0$ and $\rho_1 = \rho_2 = 0$ with the scale of the marks being driven by the timing of past IV events, (ψ_1 , $\psi_2 > 0$ and $\rho_1 = \rho_2 = 0$) and the dynamic introduced by the arrival times of the extreme events in IV ($\kappa_{12} > 0$).

Table 4 reports estimation results for all four bivariate models. In all markets, Model 3 is found to provide the best fit to the data, where the ground intensities of extreme return (λ_g^1) and IV (λ_g^2) events are driven by the size of past return marks and the timing of past return and IV events, and the scale is driven by the size of past return marks. The impact of the timing of past IV events on the intensities is evident in the positive estimates of ϕ_2 which are significant in four of the five markets. The degree of self, or cross-excitation is reflected in the combination of ϑ , ψ or ρ , and ϕ coefficients. Significant estimates of ϑ_{11} , ϕ_1 and ψ_1 for Model 3 reveal strong self-excitation in the return events with a similar pattern evident for IV events in terms of ϑ_{22} and ϕ_2 . In terms of cross excitation the results are varied, estimates of ϕ_1 and ϕ_2 are nearly always significant with estimates of ϑ_{12} and ϑ_{21} being somewhat mixed. There appears to be bi-directional cross-excitation in the DAX and Nikkei markets, with excitation from returns to IV in both the S&P500 and Nasdaq markets.

4.2 Forecasting risk

In this section, results of the tests for VaR accuracy discussed in Section 3 are presented. These backtesting results are based on the period 2012-2013. Model estimation for forecasting purposes is initially based on the in-sample period ending 30 December 2011, and then on a recursive estimation window where the models are re-estimated every week moving through the 2012-2013 period.

Before moving to a formal analysis of VaR accuracy, Figures 2 and 3 show VaR estimates and predictions at a significance level of 0.99, along with returns for the in- and out-of-sample (also with exceptions) periods respectively. Results are shown for the S&P 500 index for a selection of models across the different classes of models considered here, EGARCH + IV, BEGE+IV, univariate and bivariate Hawkes-POT (both Model 1). Beginning with Figure 2, it is clear that all the VaR estimates broadly follow the volatility of the overall market. Two observations emerge, the EGARCH+IV estimates appear to be somewhat more variable for much much of the period and both Hawkes based VaR estimates adapt to a higher level during the height of the market volatility in 2009. The lower panels in Figure 2 show the VaR estimates and associated returns during a number of important periods of crisis and heightened market volatility. It is evident that focusing in on these periods of interest highlights that the VaR estimates generated by both MPP models are less variable, certainly in comparison to those from the EGARCH+IV model. They do however adapt to noticeably higher levels during the



Figure 2: Plots of in-sample VaR estimates and returns (the negative of log returns are shown) on the S&P 500 index. VaR estimates are shown for four models across the different classes of models considered here, EGARCH + IV, BEGE+IV, univariate and bivariate Hawkes-POT. The top panel shows the full in-sample period, while the lower panels highlight various subperiods of interest.

peak of historically high volatility in 2009. Figure 3 shows the corresponding VaR predictions during the backtesting period, 2012-2013. While all the forecasts vary with the overall volatility in returns, once again, the VaR forecasts from both PP models are less variable than the EGARCH and BEGE equivalents. The exceptions from each model, BEGE+IV (×), univariate Hawkes (+) and bivariate Hawkes (∇) are also shown, with EGARCH+IV model producing no exceptions in this case. Visually speaking, there is no obvious clustering in the exceptions, it is clear the EGARCH+IV (and to some extent BEGE+IV) are not producing enough exceptions at $\alpha = 0.99$ and hence generating slightly conservative VaR predictions.

To begin the formal analysis, Table 5 reports results for the in-sample tests of VaR accuracy at $\alpha = 0.95$, 0.99, 0.995, for all GARCH, BEGE and PP models given the full in-sample period ending in December 2011. Results in the rows denoted by Exc. show that in comparison to the GARCH models, the bivariate models tend to generate slightly fewer exceptions $(X_t > VaR_{\alpha}^t)$ for most of the series. The bivariate models (include IV) generate a similar number of rejections relative to the BEGE+IV model. Overall, in the vast majority of the cases, the tests are not rejected, indicating that the models accurately describe the in-sample behaviour of the extreme events in the context of VaR estimation. The majority of the rejections that do occur, are found



Figure 3: Plots of out-of-sample VaR prediction and returns (the negative of log returns are shown) on the S&P 500 index. VaR estimates are shown for four models across the different classes of models considered here, EGARCH + IV, BEGE+IV, univariate and bivariate Hawkes-POT. Exceptions from each model, BEGE+IV (×), univariate Hawkes (+) and bivariate Hawkes (∇) are also reported. The EGARCH+IV model produced no exceptions in this case.

with the Nikkei, particularly at the highest (0.999) level of significance. Attention now turns to forecasting.

Table 6 reports results for tests of out-of-sample VaR forecast accuracy. The results are based on 1-day ahead VaR forecasts for the final backtesting period, January 2, 2012 to December 31, 2013. The first result that stands out is the frequent rejections of the LRuc, and often MCuc tests for many of the GARCH models (irrespective of whether IV is included) for all markets except the DAX. This indicates that the GARCH models are producing inaccurate VaR forecasts as the average rate of rejection is significantly different than the given level of significance in many cases. While the BEGE models also produce a number of rejections, they are less frequent than those based on the GARCH forecasts. This improvement reflects the ability of the more flexible BEGE distribution to capture tail behaviour. Apart from a number of rejections of the LRuc and MCuc tests in the case of the S&P 500, the univariate Hawkes POT models produce few other rejections. In contrast, there are no rejections produced under the bivariate Hawkes-POT forecasts across the four models, levels of significance and the five markets considered, indicating that treating the IV events an additional MPP offers gains in forecast accuracy.

Table 7 reports the MCS results based on the asymmetric quantile loss function in 24 and a level of significance of $\alpha_M = 5\%$. Table 7 shows a * when a model is included in the final MCS at a level of confidence of 95%. The most significant result is that the bivariate models are included in the final MCS in nearly every case across all markets and VaR levels. Of these models, Models 1-3, which include both the timing and size of IV events are virtually always included in the MCS. Model 4, which only includes the timing of the IV events is excluded in the majority of cases. These results, once again support the notion that treating IV as an additional point process and considering the size and timing of these events leads to the greatest benefit in terms of forecast accuracy. The univariate Hawkes-POT models are included in the MCS in well over half the cases. Of these models, Model 2, which only includes the size of past return events, is most frequently included in the MCS. The BEGE models follow closely in terms of forecast performance and remain in the MCS in about half of the cases. Finally, the GARCH models are inferior, and are excluded from the MCS in the majority of cases.

Overall these results reveal that harnessing information from IV, when treated as its own point process is beneficial. Given the bivariate Hawkes-POT models produce the most accurate forecasts across the widest range of scenarios, forecasts that pass all the tests of adequacy applied here, indicates that information regarding the timing and size of past IV extreme events is of benefit for forecasting VaR in equity markets. The benefit of including IV in a univariate point process model, or the BEGE framework is somewhat more limited, and of little use the context of GARCH models.

While the bivariate MPP models appear to dominate at the one day horizon, the final analysis determines whether adequate VaR forecasts can generated from the PP models at longer horizons. Based on 5- and 10-day VaR forecasts using the methodology discussed earlier in Section 3, Table 8 reports results for the LRuc test based on both the univariate and bivariate PP models. Once again, p-values are reported, with the results shown in bold when a rejection at 5% is observed. At the h = 5 day horizon, the adequacy of the coverage is only rejected in 22% and 12% of the cases for the univariate and bivariate models, once again indicating the information in IV is best harness through a bivariate MPP. While unsurprisingly, the rejection rates do rise moving to the longer horizon of h = 10, the adequacy of the coverage is only rejected in a quarter of cases for the bivariate models (33% of cases for the univariate models).

In summary, the bivariate Hawkes-POT models that include IV as an additional PP produce the best performing model across the widest range of scenarios. They pass all of the individual tests of VaR forecast adequacy, they are most frequently found to be amongst the most accurate under asymmetric quantile loss, and are able to generate adequate VaR forecast in most cases at a longer one-week ahead (somewhat less at two-weeks ahead) forecast horizon.

5 Conclusion

Modelling and forecasting the occurrence of extreme events in financial markets is crucially important. While there have been many studies considering the role of implied volatility (IV) for forecasting volatility, this has not been the case when dealing with extreme events. This paper addresses how best to use IV to generate forecasts of the risk of extreme events in the form of Value-at-Risk (VaR).

The BEGE model, along with traditional GARCH models including IV as an exogenous variable, coupled with EVT form the benchmark set of models. More recent advances in VaR prediction have employed marked point process (MPP) models that treat the points as the occurrence of extreme events and marks their associated size. This paper proposes a number of novel MPP models that include IV. A number of univariate models for extreme return events are developed, where the size and timing of past return events and IV are included. In addition, novel bivariate MPP models are also proposed that move beyond simply including IV as an exogenous covariate. The second dimension in the bivariate models apart from extreme stock market losses are extreme increases in IV which are themselves treated as a second MPP.

The empirical analysis here focuses on a number of major equity market indices and their associated IV indices, where the full range of models are used to generate estimates of VaR. In terms of an in-sample explanation of extreme events in equity markets, the bivariate models satisfy all backtests of VaR adequacy, while the univariate models and the BEGE models pass most. The GARCH models produce relatively frequent rejections. A similar pattern is observed when to 1-day ahead prediction of VaR. GARCH style models that include IV generate inaccurate forecasts of VaR and fail a number of tests relating to the frequency of the VaR exceptions. Univariate MPP models and BEGE models provide more accurate forecasts though still do produce a number of rejections in backtesting. It is also shown that longer horizon VaR forecasts from the bivariate MPP models satisfy most tests. Overall, the bivariate models that include the extreme IV events produce the most accurate forecasts of VaR across the full range of levels of significance and markets. A direct comparison of VaR forecast accuracy shows that the bivariate MPP models that consider the size and timing of past IV events are found to among the most accurate in the widest range of cases. These results show that while IV is certainly of benefit for predicting extreme movements in equity returns, the framework within which it is used is important. It is shown that the novel bivariate MPP model proposed here leads to superior forecasts of extreme risk in a VaR context.

References

- Aboura, S. and N. Wagner (2014). Extreme asymmetric volatility: Vix and s&p 500. Working paper 06-2014, 41.
- Balkema, A. A. and L. De Haan (1974). Residual life time at great age. The Annals of Probability 5 (2), 792 – 804.
- Becker, R., A. Clements, and A. McClelland (2009). The jump component of s&p 500 volatility and the vix index. *Journal of Banking & Finance 33*(6), 1033–1038.
- Bekaert, G. and E. A. Engstrom, E. (2015). Bad environments, good environments: A nongaussian asymmetric volatility model. *Journal of Econometrics* 186(1), 258 – 275.
- Bekaert, G. and M. Hoerova (2014). The vix, the variance premium and stock market volatility. Journal of Econometrics 183(2), 181 – 192. Analysis of Financial Data.
- Bekaert, G. and G. Wu (2000). Asymmetric volatility and risk in equity markets. Review of Financial Studies 13(1), 1–42.
- Blair, B. J., S. H. Poon, and S. J. Taylor (2001). Forecasting s&p 100 volatility: the incremental information content of implied volatilities and high-frequency index returns. *Journal of Econometrics* 105, 5–26.
- Bollerslev, T. (1986). Generalized autoregressive conditional heteroskedasticity. Journal of Econometrics 31, 307–327.
- CBOE (2003). VIX CBOE volatility index. Chicago Board Option Exchange, White Paper, www. cboe. com.
- Chavez-Demoulin, V., A. Davison, and A. McNeil (2005). A point process approach to valueat-risk estimation. *Quantitative Finance* 5, 227–234.
- Chavez-Demoulin, V. and J. McGill (2012). High-frequency financial data modeling using hawkes processes. *Journal of Banking & Finance* 36(12), 3415–3426.
- Christoffersen, P. (1998). Evaluating interval forecasts. International economic review, 841–862.
- Cotter, J. (2007). Varying the var for unconditional and conditional environments. Journal of International Money and Finance 26(8), 1338 1354.
- Danielsson, J. and C. De Vries (2000). Value-at-risk and extreme returns. Annales d'Economie et de Statistique 60, 239–270.

- Davis, R. A., T. Mikosch, et al. (2009). The extremogram: a correlogram for extreme events. Bernoulli 15(4), 977–1009.
- Davis, R. A., T. Mikosch, and I. Cribben (2012). Towards estimating extremal serial dependence via the bootstrapped extremogram. *Journal of Econometrics* 170(1), 142–152.
- Embrechts, P., T. Liniger, and L. Lin (2011). Multivariate hawkes processes: an application to financial data. *Journal of Applied Probability* 48, 367–378.
- Engle, R. and S. Manganelli (2004). Caviar. Journal of Business and Economic Statistics 22(4), 367–381.
- Fernandez, C. and M. Steel (1998). On Bayesian Modeling of Fat Tails and Skewness. Journal of the American Statistical Association 93(441).
- Giot, P. (2005). Relationships between implied volatility indexes and stock index returns. *The Journal of Portfolio Management* 31(3), 92–100.
- Glosten, L., R. Jagannathan, and D. Runkle (1993). On the relation between the expected value and the volatility of the nominal excess return on stocks. *Journal of Finance 48*, 1779–1801.
- González-Rivera, G., T.-H. Lee, and S. Mishra (2004). Forecasting volatility: A reality check based on option pricing, utility function, value-at-risk, and predictive likelihood. *International Journal of forecasting* 20(4), 629–645.
- Hansen, P., A. Lunde, and J. Nason (2003). Choosing the best volatility models: the model confidence set approach. Oxford Bulletin of Economics and Statistics 65, 839–861.
- Hansen, P., A. Lunde, and J. Nason (2011). The model confidence set. *Econometrica* 79, 453–497.
- Herrera, R. (2013). Energy risk management through self-exciting marked point process. *Energy Economics* 38, 64–76.
- Herrera, R. and N. Gonzalez (2014). The modeling and forecasting of extreme events in electricity spot markets. *Forthcoming in International Journal of Forecasting*.
- Herrera, R. and B. Schipp (2013). Value at risk forecasts by by extreme value models in a conditional duration framework. *Journal of Empirical Finance 23*, 33–47.
- Hilal, S., S.-H. Poon, and J. Tawn (2011). Hedging the black swan: Conditional heteroskedasticity and tail dependence in s&p500 and vix. *Journal of Banking & Finance 35*(9), 2374–2387.

- Kupiec, P. H. (1995). Techniques for verifying the accuracy of risk measurement models. The Journal of Derivates 3(2).
- Lin, Y.-N. and C.-H. Chang (2010). Consistent modeling of s&p 500 and vix derivatives. *Journal* of Economic Dynamics and Control 34(11), 2302–2319.
- McNeil, A. and R. Frey (2000). Estimation of tail-related risk measures for heteroscedastic financial time series: an extreme value approach. *Journal of Empirical Finance* 7, 271–300.
- Nelson, D. (1991). Conditional heteroskedasticity in asset returns: A new approach. *Econometrica* 59, 347–370.
- Ogata, Y. (1978). The asymptotic behaviour of maximum likelihood estimators for stationary point processes. Annals of the Institute of Statistical Mathematics 30(1), 243–261.
- Peng, Y. and W. L. Ng (2012). Analysing financial contagion and asymmetric market dependence with volatility indices via copulas. *Annals of Finance* 8(1), 49–74.
- Pickands, J. (1975). Statistical inference using extreme order statistics. The Annals of Statistics, 119 – 131.
- Poon, S.-H. and C. W. J. Granger (2003). Forecasting volatility in financial markets: a review. Journal of Economic Literature 41, 478–539.
- Santos, P. A. and M. F. Alves (2012). Forecasting value-at-risk with a duration-based POT method. Mathematics and Computers in Simulation 94, 295 – 309.
- Wagner, N. and A. Szimayer (2004). Local and spillover shocks in implied market volatility: evidence for the u.s. and germany. *Research in International Business and Finance 18*(3), 237 – 251.
- Ziggel, D., T. Berens, G. N. Weiß, and D. Wied (2014). A new set of improved value-at-risk backtests. Journal of Banking & Finance 48, 29–41.

A Proofs

Proof. (Proposition 1) Using the continuous representation of a Hawkes process and setting the expected intensity $\mathbb{E}[\lambda_g(t \mid \mathcal{H}_t)] = \lambda_0 < \infty$, gives for the univariate case

$$\mathbb{E}\left[\lambda_{g}(t \mid \mathcal{H}_{t})\right] = \mathbb{E}\left[\nu + \vartheta \int_{(-\infty,t) \times \mathbb{R}^{2}_{+}} f(w,z)h\left(t-s\right)N(ds \times dw \times dz)\right]$$
$$= \nu + \vartheta \mathbb{E}\left[f(w,z)\right] \mathbb{E}\left[\int_{(-\infty,t)} h\left(t-s\right)\lambda_{g}(s \mid \mathcal{H}_{s})ds\right]$$

and by assuming $\mathbb{E}[f(w,z)] = \mu_{wz}$ and by defining $\lambda_g(s \mid \mathcal{H}_s)ds = N(ds)$ leads to

$$\mathbb{E} \left[\lambda_g(t \mid \mathcal{H}_t) \right] = \nu + \vartheta \mu_{wz} \int_{(-\infty,t)} h(t-s) \lambda_0 ds$$
$$= \nu + \vartheta \mu_{wz} \lambda_0 \int_{(0,\infty)} h(s) ds$$
$$= \nu + \vartheta \mu_{wz} \lambda_0,$$

where finally $\lambda_0 = (1 - \vartheta \mu_{wz})^{-1} \nu$ is obtained. Hence, the expectation of the ground conditional intensity is finite in the univariate case, if and only if, $0 < \vartheta \mu_{wz} < 1$.

In the bivariate model the demonstration follows the same steps. Assume that the expected intensity $\mathbb{E}\left[\lambda_g^k(t \mid \mathcal{H}_t)\right] = \lambda_0^k < \infty$, for k = 1, 2. Then, by taking the unconditional expectation in (12) leads to

$$\mathbb{E}\left[\lambda_{g}^{1}(t \mid \mathcal{H}_{t})\right] = \mathbb{E}\left[\nu_{1} + \vartheta_{11}\int_{(-\infty,t)\times\mathbb{R}_{+}}f_{1}(w)h_{1}(t-s)N_{1}(ds \times dw)\right]$$
$$+\mathbb{E}\left[+\vartheta_{12}\int_{(-\infty,t)\times\mathbb{R}_{+}}f_{1}(z)h_{2}(t-s)N_{2}(ds \times dz)\right]$$
$$\mathbb{E}\left[\lambda_{g}^{2}(t \mid \mathcal{H}_{t})\right] = \mathbb{E}\left[\nu_{2} + \vartheta_{21}\int_{(-\infty,t)\times\mathbb{R}_{+}}f_{2}(w)h_{1}(t-s)N_{1}(ds \times dw)\right]$$
$$+\mathbb{E}\left[\vartheta_{22}\int_{(-\infty,t)\times\mathbb{R}_{+}}f_{2}(z)h_{2}(t-s)N_{2}(ds \times dz)\right]$$

by assuming $\mathbb{E}\left[f_k(w)\right] = \mu_w^k$ and $\mathbb{E}\left[f_k(z)\right] = \mu_z^k$, the expectations are reduced to

$$\mathbb{E}\left[\lambda_g^1(t \mid \mathcal{H}_t)\right] = \nu_1 + \vartheta_{11}\mu_w^1 \mathbb{E}\left[\int_{(-\infty,t)} h_1\left(t-s\right)N_1(ds) + \right] + \vartheta_{12}\mu_z^1 \mathbb{E}\left[\int_{(-\infty,t)} h_2\left(t-s\right)N_1(ds)\right]$$
$$\mathbb{E}\left[\lambda_g^2(t \mid \mathcal{H}_t)\right] = \nu_2 + \vartheta_{21}\mu_w^2 \mathbb{E}\left[\int_{(-\infty,t)} h_1\left(t-s\right)N_1(ds) + \right] + \vartheta_{22}\mu_z^2 \mathbb{E}\left[\int_{(-\infty,t)} h_2\left(t-s\right)N_2(ds)\right].$$

Since the kernel functions satisfy $\int_0^\infty h_k(s) ds = 1$ and $\lambda_g^k(s \mid \mathcal{H}_s) ds = N_k(ds)$ for k = 1, 2, it is

possible to express

$$\mathbb{E}\left[\lambda_g^1(t \mid \mathcal{H}_t)\right] = \nu_1 + \vartheta_{11}\mu_w^1 \mathbb{E}\left[\int_{(0,\infty)} h_1(s)\,\lambda_g^1(s \mid \mathcal{H}_s)ds\right] + \vartheta_{12}\mu_z^1 \mathbb{E}\left[\int_{(0,\infty)} h_2(s)\,\lambda_g^2(s \mid \mathcal{H}_s)ds\right] \\ \mathbb{E}\left[\lambda_g^2(t \mid \mathcal{H}_t)\right] = \nu_2 + \vartheta_{21}\mu_w^2 \mathbb{E}\left[\int_{(0,\infty)} h_1(s)\,\lambda_g^1(s \mid \mathcal{H}_s)ds\right] + \vartheta_{22}\mu_z^2 \mathbb{E}\left[\int_{(0,\infty)} h_2(s)\,\lambda_g^2(s \mid \mathcal{H}_s)ds\right],$$

which in turn is equivalent to

$$\mathbb{E} \left[\lambda_g^1(t \mid \mathcal{H}_t) \right] = \nu_1 + \vartheta_{11} \mu_w^1 \lambda_0^1 + \vartheta_{12} \mu_z^1 \lambda_0^2$$
$$\mathbb{E} \left[\lambda_g^2(t \mid \mathcal{H}_t) \right] = \nu_2 + \vartheta_{21} \mu_w^2 \lambda_0^1 + \vartheta_{22} \mu_z^2 \lambda_0^2,$$

or in matrix representation

 $\lambda_{\mathbf{0}} = \nu + (M \circ Q) \,\lambda_{\mathbf{0}},$

where $\nu = (\nu_1, \nu_2)^T$, $M = \begin{pmatrix} \mu_w^1 & \mu_z^1 \\ \mu_w^2 & \mu_z^2 \end{pmatrix}$ and $Q = \begin{pmatrix} \vartheta_{11} & \vartheta_{12} \\ \vartheta_{21} & \vartheta_{22} \end{pmatrix}$. Hence the unconditional expectation of the ground intensity given by $\lambda_0 = (\mathbf{1}_2 - M \circ Q)^{-1} \nu$ exists, if and only if, the spectral radius of the matrix $M \circ Q$ is less than one.

B Figures and Tables

Data	Model		ARI	MA specifica	tion			GAF	RCH specifica	tion		Skew-t Stu	ident dist.		
Data	Woder	μ	a_1	b_1	a_2	b_2	ω	α	β	δ	γ	ψ (skew)	ν (shape)	Log. like	AIC
	GJRGARCH	3.12E-04	0.705	-0.739			8.62E-07	0.038	0.935	0.705		0.907	7.940	18126.82	-36235.64
		(9.25E-05)	(0.077)	(0.073)			(3.87E-07)	(0.003)	(0.006)	(0.161)		(0.016)	(0.784)		
	GJRGARCH+IV	2.95E-04	0.722	-0.759			3.88E-07	0.040	0.929	0.673	5.15E-08	0.906	7.729	18135.78	-36251.56
IX	ECADOU	(9.13E-05)	(0.068)	(0.063)			(2.13E-07)	(0.004)	(0.002)	(0.126)	(3.74E-08)	(0.016)	(0.725)	10106 24	20024.00
2	EGARCH	2.88E-04 (0.00E.05)	(0.005)	-0.703			-1.18E-01 (4.07E-02)	-0.090	0.987	0.119		0.910	(1.007)	18120.34	-30234.08
0	ECADOU	(8.29E-05) 2.00E-04	(0.025)	(0.024)			(4.07E-03) 2.57E+00	(0.008)	(0.000)	(0.013)	5 11E 01	(0.016)	(1.097)	19190.00	26250.92
5P5	EGARCH+IV	3.00E-04 (0.11E.05)	(0.025)	-0.889			-3.37E+00 (8.25E-02)	-0.175	(0.005)	(0.018)	0.11E-01 (9.92E-02)	(0.017)	(1.006)	18189.92	-30339.83
SS	CARCH	(9.11E-03) 4.73E-04	0.794	0.841			(8.33E-02) 5.14E-07	0.063	0.034	(0.018)	(0.0312-03)	0.000	7.005	18070.46	36124.03
	onion	(8.31E-05)	(0.047)	(0.042)			(2.61E-07)	(0.005)	(0.004)			(0.017)	(0.656)	10070.40	-30124.33
	GARCH+IV	4.68E-04	0.785	-0.832			2.76E-07	0.060	0.936		1.61E-08	0.909	6.958	18072.18	-36126.36
		(8.40E-05)	(0.049)	(0.044)			(2.17E-07)	(0.001)	(0.001)		(4.41E-08)	(0.017)	(0.618)		00020000
	GJRGARCH	3.76E-04	-0.796	-0.929	0.808	0.942	2.10E-06	0.065	0.911	0.443	(,	0.891	11.881	15173.94	-30325.88
		(1.46E-04)	(0.032)	(0.007)	(0.030)	(0.009)	(1.11E-06)	(0.011)	(0.012)	(0.072)		(0.018)	(1.761)		
	GJRGARCH+IV	3.48E-04	-0.666	-0.973	0.676	0.976	2.56E-09	0.056	0.895	0.594	2.40E-07	0.887	13.002	15192.05	-30360.10
×		(1.42E-04)	(0.004)	(0.003)	(0.001)	(0.002)	(3.83E-07)	(0.003)	(0.004)	(0.102)	(3.94E-08)	(0.018)	(1.956)		
PA	EGARCH	3.33E-04	-0.847	-0.985	0.852	0.993	-1.48E-01	-0.090	0.983	0.157		0.888	11.540	15181.30	-30340.61
5		(1.48E-04)	(0.004)	(0.002)	(0.003)	(0.001)	(8.08E-03)	(0.008)	(0.001)	(0.015)		(0.018)	(0.231)		
×	EGARCH+IV	4.02E-04	0.020	-0.895	-0.031	0.877	-3.80E+00	-0.177	0.754	0.058	5.36E-01	0.886	13.474	15251.98	-30479.97
DA:		(1.38E-04)	(0.047)	(0.044)	(0.050)	(0.048)	(6.19E-01)	(0.016)	(0.039)	(0.022)	(9.16E-02)	(0.018)	(2.113)		
П	GARCH	6.35E-04	-0.666	-0.972	0.678	0.977	1.36E-06	0.083	0.912			0.898	10.438	15134.10	-30248.20
		(1.45E-04)	(0.002)	(0.004)	(0.002)	(0.001)	(1.45E-06)	(0.017)	(0.018)			(0.017)	(1.204)		
	GARCH+IV	6.24E-04	-0.666	-0.972	0.678	0.977	6.82E-09	0.081	0.903		1.48E-07	0.898	10.775	15142.63	-30263.27
		(1.47E-04)	(0.003)	(0.004)	(0.003)	(0.001)	(3.89E-07)	(0.003)	(0.003)		(2.12E-08)	(0.018)	(1.374)		
	GJRGARCH	2.20E-04	-0.337	0.492	0.283	-0.525	1.04E-06	0.031	0.930	0.997		0.893	10.098	11175.19	-22328.38
	amathan w	(1.16E-04)	(0.179)	(0.148)	(0.168)	(0.133)	(2.50E-06)	(0.009)	(0.028)	(0.335)		(0.013)	(1.895)		
	GJRGARCH+IV	2.06E-04	-0.275	0.552	0.220	-0.586	2.00E-08	0.032	0.921	0.958	1.02E-07	0.889	10.801	11186.98	-22349.97
0	DOLDON	(1.29E-04)	(0.265)	(0.244)	(0.254)	(0.228)	(2.73E-07)	(0.006)	(0.003)	(0.198)	(2.59E-08)	(0.020)	(1.601)		
IX.	EGARCH	1.33E-04	0.057	-0.542	-0.101	0.527	-1.32E-01	-0.116	0.986	0.102		0.889	10.231	11183.27	-22344.54
-	ECADOU	(1.19E-04)	(0.029)	(0.051)	(0.029)	(0.052)	(1.98E-03)	(0.009)	(0.002)	(0.003)	1.0012.01	(0.021)	(1.482)	11000 70	00400 51
F	EGARCH+IV	1.18E-04 (1.95E-04)	(0.070)	-0.301	-0.152	(0.072)	-1.50E+00	-0.182	(0.901	(0.000)	1.98E-01 (0.74E-02)	(0.021)	12.840	11220.70	-22429.31
Ω	CADCII	(1.25E-04)	(0.070)	(0.073)	(0.070)	(0.073)	(2.81E-02) 0.14E-07	(0.013)	(0.002)	(0.009)	(8.74E-03)	(0.021)	(2.498)	11105-27	00020 74
	GARCH	4.55E-04 (1.10E.04)	-0.127	(0.110)	(0.154)	-0.702	9.14E-07 (7.00E-07)	0.078	(0.011)			0.900	0.104	11120.07	-22230.74
	CARCHIN	(1.19E-04) 4 20E 04	(0.159)	0.661	(0.134)	0.608	(7.99E-07) 2.68E-07	0.020	0.000		1.15E-07	(0.021)	(1.120)	11129-19	00040.05
	GARCHTIV	(1.22E_04)	(0.142)	(0.110)	(0.137)	-0.098	(3.70E-07)	(0.003)	(0.004)		(1.06E-08)	(0.021)	(1.107)	11132.12	-22242.20
	CIRCARCH	0.72E-06	0.608	0.714	(0.157)	(0.100)	4.64E-06	0.073	0.806	0.361	(1.0012-00)	0.022	13 100	0833.40	106/18/08
	OSITOAITON	(2.24E-04)	(0.097)	(0.072)			(2.39E-06)	(7.56E-03)	(6 72E-03)	(5.97E-02)		(0.023)	(2.319)	3000.43	-13040.30
	GIRGARCH+IV	-3.26E-05	0.611	-0.628			2.83E-09	0.057	0.865	0.540	4.92E-07	0.923	13.950	9851-19	-19682-38
	051t0/ittofi+iv	(2.00E-04)	(0.661)	(0.651)			(7.67E-07)	(4.02E-03)	(7 50E-03)	(1 10E-01)	(9.64E-08)	(0.022)	(2.727)	3001.13	-13002.30
CX .	EGABCH	-7.46E-05	0.312	-0.335			-2.41E-01	-0.088	0.972	0.160	(0.0111-00)	0.925	12.960	9838.55	-19659.11
>	Bonnon	(2.56E-04)	(0.033)	(0.033)			(5.92E-02)	(1.07E-02)	(6.99E-03)	(8.17E-02)		(0.024)	(3.274)	0000.00	10000.11
.e	EGARCH+IV	5.52E-05	0.770	-0.782			-6.76E+00	-0.178	0.556	-0.006	9.14E-01	0.940	14.052	9884.52	-19749.04
ikk		(2.06E-04)	(0.276)	(0.270)			(1.41E+00)	(2.37E-02)	(8.99E-02)	(4.29E-02)	(1.99E-01)	(0.023)	(2.929)		
Z	GARCH	2.42E-04	0.797	-0.824			3.44E-06	0.084	0.902	()	()	0.921	11.179	9810.78	-19605.56
		(1.89E-04)	(0.141)	(0.132)			(1.81E-06)	(1.27E-02)	(1.45E-02)			(0.023)	(1.891)		
	GARCH+IV	1.96E-04	0.788	-0.814			1.13E-08	0.077	0.872		4.19E-07	0.920	12.075	9825.88	-19633.75
		(1.91E-04)	(0.144)	(0.136)			(7.87E-07)	(5.27E-03)	(6.75E-03)		(8.66E-08)	(0.023)	(2.138)		
	GJRGARCH	1.66E-04	-0.001	-0.913	-0.011	0.891	1.23E-06	0.026	0.942	0.982		0.873	17.846	7945.15	-15868.30
		(1.71E-04)	(0.068)	(0.048)	(0.076)	(0.053)	(2.39E-06)	(0.004)	(0.021)	(0.380)		(0.023)	(0.653)		
	GJRGARCH+IV	1.56E-04	-0.007	-0.477	-0.027	0.440	6.79E-07	0.033	0.930	0.819	7.47E-08	0.867	15.936	7948.68	-15873.37
Z		(1.97E-04)	(0.304)	(0.389)	(0.307)	(0.405)	(4.27E-07)	(0.007)	(0.003)	(0.243)	(2.49E-08)	(0.023)	(3.642)		
XX	EGARCH	9.74E-05	0.020	-0.922	-0.035	0.904	-8.99E-02	-0.091	0.990	0.091		0.863	16.249	7945.70	-15869.39
<u></u>		(1.42E-04)	(0.017)	(0.025)	(0.019)	(0.028)	(1.78E-03)	(0.009)	(0.001)	(0.003)		(0.023)	(4.190)		
dae	EGARCH+IV	1.16E-04	-0.192	-0.394	0.167	0.356	-1.46E+00	-0.164	0.899	0.046	1.76E-01	0.849	20.539	7971.87	-15919.75
7 as	aupon	(1.95E-04)	(0.150)	(0.072)	(0.152)	(0.073)	(2.69E-02)	(0.014)	(0.001)	(0.011)	(8.28E-03)	(0.023)	(6.307)		
~	GARCH	3.59E-04	1.605	-0.608	-1.653	0.654	1.17E-06	0.069	0.927			0.880	15.341	7917.08	-15814.16
	CADOLL IV	(8.16E-05)	(0.001)	(0.001)	(0.002)	(0.002)	(6.83E-07)	(0.002)	(0.002)		6.0.1E.00	(0.024)	(4.224)	2 010 14	15004.00
	GARCH+IV	4.74E-04 (1.84E-04)	(0.208)	0.470	-0.092	-0.513	4.62E-07	0.067	0.924		0.04E-08	0.880	12.918	7913.14	-15804.29
		(1.84E-04)	(0.528)	(0.259)	(0.324)	(0.263)	(0.05E-07)	(0.003)	(0.004)		(3.91E-08)	(0.024)	(2.321)		
Data	Model	<i>p</i> ₀	σ_p	ρ_p	ϕ_p^{\top}	ϕ_p^-	n ₀	σ _n	ρ_n	ϕ_n^{+}	ϕ_n^-			101/0.05	00000 51
S&P500 - VIX	BEGE	0.026	0.002	0.970	0.006	0.043	0.096	0.005	0.862	-0.034	0.274			18140.25	-36260.51
	DECE IN	(0.020)	(5.0.E-06)	(2.2.E-04)	(0.005) 5.6 E 02	(0.007)	(0.021)	(0.1.E-05)	(5.2.E-04)	(0.021)	(0.037)	0.040		10140 66	26077 20
	DEGE+IV	(2.1 E.06)	(2.4 E.06)	(2.2 F 05)	5.0.E-05 (5.2.E.07)	(1.4 E.06)	(1.6 E.05)	(2.1 E.06)	(26 E 04)	-0.037 (1.2 E.05)	(4.4 E 05)	(0.8 E 07)		18149.00	-30211.32
DAY VDAY	DECE	(3.1.12-00)	(3.4.12-00)	(3.2.12-03)	(0.2.12-07)	(1.4.12-00)	(1.0.12-03)	(2.1.15-00)	(3.0.12-04)	(1.2.12-00)	(4.4.12-00)	(9.8.12-07)		15101 00	20202.22
DAA - VDAA	DEGE	(0.090	(1.1.E.0.4)	0.958	(0.012)	(0.028	(0.012)	(4.8 E.04)	(0.011)	-0.055	(0.002)			13101.00	-30303.33
	DECE IN	(0.021)	(1.1.E-04)	(2.0.E-04)	(0.013)	(0.009)	(0.018)	(4.8.E-04)	(0.011)	(0.009)	(0.025)	0.040		15164.00	20200.05
	DEGE+IV	(0.004)	(2.0 E 10)	(8 0 E 05)	(2.2 E 0.4)	(0.001)	(0.007)	(2.4 E.04)	(2.0 E.05)	-0.055	(0.012)	(0.002)		15104.92	-30309.85
DIL VVD	DECE	0.019	(3.0.12-10)	(8.0.12-03)	(3.2.E-04) 4.2 E 06	0.062	0.101	(3.4.15-04)	(2.9.12-00)	0.046	0.248	(0.003)		11177.56	00225-12
201 - VAD	20202	(0.041)	(5.7 E.04)	(10E.04)	(0.013)	(0.052)	(0.022)	(7.4 E-04)	(7.1 E.05)	(0.032)	(0.001)			11111.00	22000.10
	BEGE+IV	0.024	0.002	0.961	3.0 E-06	0.059	0.104	0.004	0.893	-0.050	0.245	0.056		11179 66	-99337 33
	DECETIV	(3.7.F-04)	(5.7.E-07)	(1.0.E-05)	(1.1.E-00)	(2.5.F-04)	(1.9 E-05)	(2.9.E-06)	(3.4.E-04)	(0.002)	(0.012)	(0.007)		111/9.00	22001.00
Nikkei - VXI	BEGE	0.030	0.004	0.952	0.051	0.034	0.123	0.007	0.851	-0.021	0.247	(0.001)		9827 51	-19635.02
- (manor = 1770)	5101	(0.009)	(4.5.E-04)	(1.8.E-04)	(0.016)	(0.018)	(0.015)	(0,001)	(2.0,E-04)	(0.017)	(0.046)			0021.01	10000.02
	BEGE+IV	0.031	0,004	0.951	0.055	0.024	0.167	0,006	0.844	-0.023	0.254	0.228		9832.16	-19642.32
		(0.011)	(2.0.E-04)	(0.011)	(0.012)	(0.005)	(0.040)	(0.001)	(0.003)	(0.005)	(0.022)	(0.159)		0.000000	
Nasdaq - VXN	BEGE	7.5.E-06	0.002	0.973	6.0.E-06	0.038	0.144	0.005	0.897	-0.044	0.240	()		7951.47	-15882.95
1		(0.015)	3.5.E-04	(0.011)	(0.013)	(0.013)	(0.088)	(4.8.E-04)	(0.059)	(0.015)	(0.094)				
	BEGE+IV	5.1.E-06	0.003	0.974	-1.0.E-06	0.020	0.076	0.005	0.927	-0.080	0.217	0.549		7955.32	-15888.64
		(0.010)	(0.001)	(0.008)	(0.011)	(0.019)	(0.012)	(9.7.E-06)	(8.9.E-05)	(0.019)	(0.023)	(0.351)			

Table 2: Estimation results for the volatility models (GARCH and BEGE) applied to the returns on each of the equity indices, based on the full in-sample period ending 30 December 2011. Standard errors based on inversion of the numerical Hessian are reported in parentheses. Log.like corresponds to the log-likelihood value obtained, while the AIC is the Akaike Information Criterion.

Data	Model	ν	θ	ϕ	ψ	ρ	κ_0	κ_1	ξ	Log. like	AIC
×	1	0.031	0.386	0.046	27.660	5.351	0.004	0.017	-0.097	446.904	-877.807
VL		(0.005)	(0.049)	(0.008)	(3.408)	(2.006)	(0.000)	(0.002)	(0.034)		
0	2	0.033	0.449	0.054	32.389		0.004	0.019	-0.092	443.718	-873.435
250		(0.005)	(0.050)	(0.008)	(2.944)		(0.000)	(0.002)	(0.035)		
&F	3	0.021	0.794	0.038			0.004	0.030	0.043	410.805	-809.609
∞		(0.004)	(0.054)	(0.006)			(0.000)	(0.003)	(0.038)		
	1	0.040	0.322	0.071	36.720	1.374	0.005	0.016	-0.177	276.613	-537.225
Ð		(0.007)	(0.060)	(0.012)	(4.567)	(1.007)	(0.001)	(0.003)	(0.044)		
5	2	0.041	0.358	0.074	38.681		0.005	0.018	-0.173	275.746	-537.493
- I		(0.006)	(0.058)	(0.012)	(4.336)		(0.001)	(0.003)	(0.044)		
DJ	3	0.024	0.761	0.048			0.004	0.033	-0.025	252.712	-493.423
		(0.006)	(0.069)	(0.009)			(0.001)	(0.005)	(0.050)		
×	1	0.044	0.259	0.073	43.010	1.655	0.005	0.016	-0.179	333.364	-650.727
ΥC		(0.005)	(0.046)	(0.010)	(4.257)	(1.026)	(0.001)	(0.003)	(0.033)		
ΓΛ	2	0.044	0.294	0.074	43.319		0.005	0.018	-0.180	332.085	-650.170
, ,		(0.005)	(0.045)	(0.010)	(4.076)		(0.001)	(0.003)	(0.034)		
Ϋ́	3	0.027	0.736	0.050			0.004	0.042	-0.034	292.604	-573.208
Д		(0.005)	(0.057)	(0.007)			(0.001)	(0.004)	(0.035)		
ſ	1	0.060	0.164	0.066	22.564	2.927	0.005	0.023	-0.101	150.623	-285.246
XX		(0.009)	(0.067)	(0.012)	(5.068)	(0.906)	(0.001)	(0.004)	(0.050)		
	2	0.060	0.207	0.078	31.455		0.004	0.025	-0.029	146.767	-279.534
kei		(0.009)	(0.072)	(0.016)	(4.118)		(0.001)	(0.004)	(0.040)		
Vilk	3	0.034	0.630	0.047			0.004	0.038	0.133	121.026	-230.052
<u> </u>		(0.008)	(0.093)	(0.012)			(0.001)	(0.007)	(0.059)		
Z	1	0.037	0.386	0.059	35.745	0.290	0.006	0.019	-0.231	261.475	-506.949
XX		(0.005)	(0.058)	(0.010)	(4.484)	(1.044)	(0.001)	(0.003)	(0.052)		
<u>_</u>	2	0.037	0.389	0.060	36.398		0.006	0.019	-0.230	261.436	-508.871
lag		(0.006)	(0.054)	(0.010)	(3.658)		(0.001)	(0.003)	(0.044)		
lasc	3	0.024	0.789	0.047			0.004	0.039	-0.056	232.970	-453.939
Z		(0.006)	(0.065)	(0.007)			(0.001)	(0.005)	(0.043)		

Table 3: Estimation results for the univariate Hawkes-POT models based on extreme events in negative log-returns in stock market indices and positive changes in IV indices, ending 30 December 2011. Standard errors based on inversion of the numerical Hessian are reported in parentheses. Log.like corresponds to the log-likelihood of the model. AIC is the Akaike Information Criterion.

Data	Model	ν_1	ϑ_{11}	ϑ_{12}	ϕ_1	ψ_1	ψ_2	ρ_1	ρ_2	ν_2	ϑ_{21}	ϑ_{22}	ϕ_2	ĸo	κ_1	κ_{12}	ξ	log. like	AIC
~	1	0.033	0.448	0.000	0.054	32.417	2.732	0.000	65.385	0.066	0.006	0.323	0.030	0.004	0.019	0.000	-0.093	-1460.58	2953.15
9		(0.005)	(0.043)	(0.008)	(0.001)	(0.002)	(0.034)	(0.003)	(0, 010)	(0.005)	(0.108)	(0.013)	(0.001)	(2.009)	(5.156)	(1, 336)	(0.002)		
~	9	0.024	0.445	0.000	0.054	22.647	0.000	0.000	22.647	0.065	0.025	0.218	0.021	0.004	0.010	0.000	0.002)	1461.07	2050-14
8	4	(0.005)	(0.050)	(0.000)	(0.009	(0.022)	(0.000)	(0.000)	(0.022)	(0.010)	(0.025	(0.110)	(0.014)	(0.004	(0.000)	(0.000)	(0.027)	-1401.07	2550.14
Ë		(0.005)	(0.050)	(0.003)	(0.008)	(2.933)	(0.002)	(0.002)	(2.933)	(0.010)	(0.017)	(0.118)	(0.014)	(0.000)	(0.002)	(0.000)	(0.035)		
28	3	0.034	0.445	0.000	0.054	32.647			32.647	0.065	0.025	0.318	0.031	0.004	0.019		-0.093	-1461.07	2946.14
		(0.005)	(0.050)	(0.003)	(0.008)	(2.933)			(2.933)	(0.010)	(0.017)	(0.118)	(0.014)	(0.000)	(0.002)		(0.035)		
	4	0.020	0.803	0.000	0.035					0.063	0.000	0.378	0.029	0.003	0.023	0.019	0.035	-1491.98	3007.96
		(0.004)	(0.055)	(0.005)	(0.006)					(0.010)	(0.002)	(0.099)	(0.010)	(0.001)	(0.005)	(0.009)	(0.037)		
	1	0.036	0.360	0.000	0.075	38.658	0.121	0.000	0.624	0.050	0.000	0.447	0.033	0.005	0.020	0.000	-0.166	-920.33	1872.65
р		(0.006)	(0.050)	(0.001)	(0.000)	(1.740)	(0.051)	(0.002)	(0.315)	(0.011)	(0.000)	(0.114)	(0.017)	(0.001)	(0.002)	(0.000)	(0.047)		
×.	9	0.026	0.260	0.000	0.075	29.659	0.000	0.000	28.658	0.050	0.000	0.447	0.022	0.005	0.020	0.000	0.164	020.22	1868.65
<u></u>	2	(0.006)	(0.050)	(0.004)	(0.012)	(4.457)	(0.002)	(0.002)	(4.457)	(0.011)	(0.001)	(0.125)	(0.010)	(0.001)	(0.002)	(0.000)	(0.048)	-520.00	1000.00
5	0	(0.000)	(0.009)	(0.004)	(0.013)	(4.407)	(0.002)	(0.002)	(4.457)	(0.011)	(0.001)	(0.125)	(0.019)	(0.001)	(0.003)	(0.000)	(0.040)	000.00	1001.05
Ц	3	0.036	0.300	0.000	0.075	38.008			38.038	0.050	0.000	0.447	0.033	0.005	0.020		-0.104	-920.33	1804.00
		(0.006)	(0.059)	(0.004)	(0.013)	(4.457)			(4.457)	(0.011)	(0.001)	(0.125)	(0.019)	(0.001)	(0.003)		(0.048)		
	4	0.021	0.762	0.009	0.045					0.053	0.000	0.420	0.040	0.003	0.033	0.010	-0.005	-941.05	1906.11
		(0.008)	(0.074)	(0.092)	(0.009)					(0.010)	(0.001)	(0.112)	(0.019)	(0.001)	(0.006)	(0.008)	(0.052)		
×	1	0.000	0.183	0.642	0.094	49.216	0.000	0.000	102.345	0.040	0.002	0.610	0.004	0.006	0.015	0.000	-0.191	-1394.83	2821.65
-AC		(0.007)	(0.040)	(0.110)	(0.013)	(4.202)	(0.004)	(0.004)	(14.201)	(0.011)	(0.015)	(0.126)	(0.001)	(0.001)	(0.002)	(0.001)	(0.034)		
5	2	0.000	0.180	0.643	0.094	49 785	0.002	0.002	49 785	0.041	0.023	0.577	0.004	0.006	0.015	0.000	-0.192	-1439 64	2907 29
1.1	-	(0.007)	(0.030)	(0.070)	(0.013)	(4 503)	(0.001)	(0.001)	(4 593)	(0.010)	(0.014)	(0.003)	(0.001)	(0.001)	(0.002)	(0.000)	(0.034)		
×	2	0.000	0.180	0.644	0.004	40.785	(0.001)	(0.001)	40.785	0.041	0.022	0.577	0.004	0.006	0.015	(0.000)	0.102	1205 74	9815.48
ã	0	(0.007)	(0.040)	(0.110)	(0.012)	(4 COA)			(4.004)	(0.011)	(0.015)	(0.100)	(0.001)	(0.001)	(0.000)		(0.024)	-1050.14	2010.40
		(0.007)	(0.040)	(0.110)	(0.013)	(4.004)			(4.004)	(0.011)	(0.013)	(0.120)	(0.001)	(0.001)	(0.002)	0.017	(0.034)	1 100 01	0000 00
	4	0.017	0.699	0.137	0.042					0.051	0.001	0.493	0.039	0.003	0.037	0.017	-0.055	-1439.64	2903.29
		(0.007)	(0.068)	(0.089)	(0.009)					(0.010)	(0.061)	(0.116)	(0.014)	(0.001)	(0.005)	(0.008)	(0.039)		
D	1	0.057	0.181	0.062	0.079	32.442	0.447	0.000	0.681	0.065	0.000	0.354	0.063	0.005	0.024	0.000	-0.031	-1059.93	2151.85
×.		(0.009)	(0.073)	(0.001)	(0.016)	(4.117)	(72.437)	(4.841)	(128.903)	(0.012)	(0.001)	(0.236)	(0.002)	(0.002)	(0.004)	(0.081)	(0.040)		
	2	0.057	0.168	0.068	0.080	33.449	0.003	0.003	33.449	0.066	0.019	0.318	0.066	0.005	0.023	0.000	-0.033	-1059.58	2147.16
čei		(0.010)	(0.079)	(0.034)	(0.017)	(4.860)	(0.000)	(0.000)	(4.860)	(0.009)	(0.013)	(0.096)	(0.024)	(0.001)	(0.004)	(0.001)	(0.040)		
ik	3	0.057	0.168	0.068	0.080	33.449	()	()	33.449	0.066	0.019	0.318	0.066	0.005	0.023	()	-0.033	-1059.58	2143.16
Z		(0.010)	(0.079)	(0.034)	(0.017)	(4.860)			(4.860)	(0.009)	(0.013)	(0.096)	(0.024)	(0.001)	(0.004)		(0.040)		
	4	0.012	0.719	0.125	0.021	(1.000)			(1.000)	0.068	0.000	0.222	0.082	0.002	0.022	0.020	0.112	1076.00	2177.08
	-1	(0.010)	(0.007)	(0.000)	(0.021					(0.000)	(0.001)	(0.022)	(0.032	(0.003	(0.022	(0.000)	(0.055)	-1070.55	2111.50
		(0.010)	(0.097)	(0.069)	(0.007)					(0.008)	(0.001)	(0.079)	(0.019)	(0.001)	(0.008)	(0.008)	(0.055)		
	-																		
2	1	0.026	0.423	0.000	0.065	35.420	9.092	0.000	7.960	0.028	0.000	0.762	0.008	0.005	0.022	0.002	-0.247	-801.20	1634.40
5		(0.030)	(0.027)	(0.015)	(0.006)	(0.024)	(0.040)	(0.021)	(0.023)	(0.036)	(0.016)	(0.065)	(0.014)	(0.021)	(0.033)	(0.026)	(0.029)		
<u> </u>	2	0.037	0.388	0.000	0.060	36.490	0.000	0.000	36.490	0.032	0.004	0.742	0.007	0.006	0.019	0.000	-0.231	-801.22	1630.45
lag		(0.006)	(0.055)	(0.001)	(0.010)	(3.684)	(0.002)	(0.002)	(3.684)	(0.013)	(0.023)	(0.119)	(0.006)	(0.001)	(0.003)	(0.000)	(0.044)		
asc	3	0.037	0.388	0.000	0.060	36.494			36.494	0.032	0.005	0.743	0.007	0.006	0.019		-0.231	-800.50	1624.99
Z		(0.006)	(0.055)	(0.001)	(0.010)	(3.684)			(3.684)	(0.013)	(0.023)	(0.119)	(0.006)	(0.001)	(0.003)		(0.044)		
	4	0.024	0.788	0.000	0.047	((-)0-)	0.033	0.000	0.739	0.008	0.004	0.038	0.005	-0.059	-821.64	1667.29
		(0.006)	(0.065)	(0.001)	(0.007)					(0.011)	(0.001)	(0.115)	(0.005)	(0.001)	(0.005)	(0.008)	(0.042)	021.01	
		(0.000)	(0.000)	(0.001)	(0.007)					(0.011)	(0.001)	(0.110)	(0.003)	(0.001)	(0.000)	(0.008)	(0.040)		

Table 4: Estimation results for the bivariate Hawkes-POT models based extreme events in negative log-returns in stock indices and positive level-changes in IV indices, ending 30 December 2011. Standard errors based on inversion of the numerical Hessian are reported in parentheses. log.like corresponds to the log-likelihood of the model. AIC is the Akaike Information Criterion.

Dotumo	Chatiotico				V	olatility Model	s (In-sampl	e)									t	Univaria	te Hawkes	-POT (In	sample)						Bivariate	Hawkes	-POT (.	In-samp	əle)			
neturns	Statistics	GJRGARCH	GJRGARCH+IV	EGARCH	EGA	RCH+IV	GARO	CH	GARC	H+IV	1	BEGE		BEGE+	IV		M1		M	2		M3			M1		M2			M3			M4	
	a loval	0.05 0.00 0.000	0.05 0.00 0.000 0	0.05 0.00 0.0	000 0.05	0.00 0.000	0.05 0.00	0.000	0.05 0.0	0 0 0 0 0	0.95	0.00 (000 0	05 0.00	0.000	0.05	0.00	0.000	0.05 0.0	0 0 0 0 0	0.05	0.00	0.000	0.05	0.00 0	000 0.0	5 0.00	0.000	0.05	0.00	0.000	0.05	0.00 0	1 000
	a- ievei	0.35 0.33 0.333	0.30 0.33 0.333 0	0.30 0.33 0.3	0.30	0.33 0.333	0.30 0.33	0.000	0.30 0.3	0.000	0.30	0.00 (3.333 0.	30 0.33	0.333	0.30	0.33	0.000	0.30 0.2	0.000	0.30	0.33	0.333	0.30	0.33 0		0 0.33	0.000	0.30	0.00	0.000	0.30	5.55 0.	
	Exc.	286 54 7	285 51 7	280 57	9 293	60 7	297 61	10	297 61	. 10	225	39	10 24	47 43	10	245	52	8	259 49	9 10	260	52	11	260	49	8 24	6 52	8	259	49	8	243	53	9
	LRuc	0.59 0.84 0.55	0.63 0.54 0.55 0	0.87 0.84 0.	.18 0.34	0.54 0.55	0.23 0.46	0.09	0.23 0.4	6 0.09	0.00	0.02	0.09 0.	06 0.08	0.09	0.04	0.64	0.33	0.26 0.3	8 0.09	0.28	0.64	0.04	0.28	0.38 (0.33 0.0	15 0.64	0.33	0.26	0.38	0.33	0.03	0.74 C	0.18
	LRind	0.74 0.30 0.89	0.71 0.33 0.89 (0.44 0.62 0.	86 0.10	0.25 0.89	0.30 0.70	0.85	0.59 0.7	0 0.85	0.45	0.46	0.85 0.	18 0.41	0.85	0.50	0.32	0.88	0.40 0.3	5 0.85	0.17	0.52	0.83	0.17	0.35 (1.88 0.3	5 0.32	0.88	0.40	0.35	0.88	0.47	0.31 (0.86
	I D oo	0.92 0.52 0.92	0.84 0.52 0.82	0.72 0.86 0	40 0.16	0.42 0.92	0.98 0.71	0.92	0.42 0.7	1 0.92	0.00	0.05	0.92 0	07 0.15	0.92	0.10	0.55	0.61	0.27 0.4	4 0.92	0.99	0.72	0.19	0.99	0.44 0	61 0/	0 0.55	0.61	0.27	0.44	0.61	0.09	0.57 (0.40
	LACC	0.82 0.38 0.83	0.84 0.32 0.83	0.75 0.80 0.	.40 0.10	0.45 0.65	0.28 0.71	0.25	0.42 0.7	1 0.23	0.00	0.03	0.23 0.	07 0.13	0.25	0.10	0.00	0.01	0.57 0.4	4 0.25	0.22	0.15	0.12	0.22	0.44 (.01 0.0	9 0.00	0.01	0.57	0.44	0.01	0.08	5.57 0	J.40
S&P500 -VIX	DQhit	0.74 0.31 0.89	0.72 0.33 0.89 0	0.45 0.62 0.	.86 0.11	0.25 0.89	0.31 0.71	0.85	0.60 0.7	1 0.85	0.46	0.46	0.85 0.	19 0.41	0.85	0.51	0.32	0.88	0.41 0.3	5 0.85	0.18	0.52	0.83	0.18	0.35 (0.88 0.3	6 0.32	0.88	0.41	0.35	0.88	0.48	0.31 0	J.86
	DQVaR	0.62 0.48 0.99	0.73 0.52 0.99 0	0.74 0.65 0.	.99 0.27	0.42 0.99	0.47 0.66	0.98	0.68 0.6	7 0.98	0.03	0.08	0.00 0.	05 0.02	0.00	0.03	0.51	0.99	0.53 0.5	3 0.98	0.31	0.61	0.98	0.39	0.52 (.99 0.0	02 0.51	0.99	0.65	0.53	0.99	0.04	0.50 C	0.99
	MCuc	0.59 0.90 0.51	0.64 0.54 0.43	0.89 0.85 0.	16 0.34	0.50 0.41	0.23 0.46	0.06	0.22 0.4	6 0.08	0.00	0.02	0.06 0.	06 0.10	0.08	0.05	0.63	0.25	0.24 0.3	7 0.08	0.29	0.63	0.04	0.28	0.37 (.35 0.0	0.64	0.35	0.26	0.43	0.35	0.04	0.80 0	0.16
	MCL	0.00 0.00 0.00	0.02 0.20 0.00	0.70 0.00 0	40 0.10	0.22 0.20	0.10 0.15	0.04	0.10 0.1	0.07	0.00	0.02	0.14 0	CO 0.01	0.14	0.00	0.01	0.20	0.00 0.0	r 0.20	0.00	0.00	0.01	0.00	0.07 (20 04	0.00	0.20	0.00	0.07	0.20	0.00	0.01 (0.00
	MCind	0.88 0.28 0.62	0.93 0.32 0.62 0	0.78 0.06 0.	.42 0.12	0.33 0.32	0.19 0.15	0.24	0.18 0.1	8 0.25	0.20	0.03	0.14 0.	08 0.01	0.14	0.00	0.01	0.30	0.00 0.0	5 0.52	0.00	0.00	0.21	0.00	0.07 (.30 0.0	0.00	0.30	0.00	0.05	0.30	0.00	J.01 0	0.22
	MCcc	0.24 0.58 0.73	0.15 0.63 0.74 0	0.46 0.09 0.	.86 0.25	0.66 0.66	0.38 0.32	0.49	0.37 0.3	6 0.50	0.38	0.07	0.29 0.	66 0.03	0.29	0.00	0.02	0.62	0.01 0.1	0 0.62	0.01	0.13	0.41	0.01	0.14 (1.60 U.	0 0.13	0.60	0.01	0.10	0.60	0.01	J.02 0	J.44
	Exc.	292 58 6	284 58 6	294 58	10 297	53 8	318 53	9	309 54	8	216	34	3 20	07 31	3	253	60	15	254 60) 13	229	43	16	248	62	17 24	8 58	16	246	61	5	232	43	6
	LRuc	0.40 0.76 0.86	0.73 0.76 0.86	0.34 0.76 0.	.09 0.26	0.72 0.33	0.02 0.72	0.18	0.06 0.8	2 0.33	0.01	0.01	0.32 0.	00.00	0.32	0.99	0.20	0.00	0.94 0.2	0.00	0.12	0.27	0.00	0.75	0.12 0	.00 0.1	5 0.31	0.00	0.65	0.15	0.98	0.17	0.27 (0.68
	I Rind	0.00 0.02 0.01	0.35 0.02 0.01 0	0.02 0.02 0	85 0.10	0.11 0.88	0.07 0.11	0.86	0.06 0.1	1 0.88	0.11	0.50	0.05 0	08 0.54	0.05	0.13	0.74	0.77	0.14 0.5	4 0.80	0.16	0.06	0.75	0.05	0.70 (73 0	1 0.70	0.75	0.37	0.99	0.02	0.67	0.38 (0.00
	LIGHIG	0.03 0.02 0.31	0.00 0.02 0.01 0	0.02 0.02 0.	04 0.00	0.07 0.00	0.01 0.11	0.00	0.00 0.1	0.00	0.01	0.00	0.50 0.	00 0.01	0.00	0.10	0.14	0.00	0.14 0.7	1 0.00	0.10	0.00	0.10	0.00	0.00 0	.10 0.	0.70	0.10	0.01	0.17	0.02	0.01	0.97 (0.01
	LACC	0.17 0.08 0.98	0.01 0.08 0.98 0	J.U 4 0.08 0.	.24 0.22	0.23 0.02	0.01 0.23	0.40	0.03 0.2	8 0.02	0.01	0.04	0.01 0.	00 0.01	0.01	0.55	0.41	0.00	0.34 0.4	1 0.01	0.11	0.09	0.00	0.13	0.29 0	.00 0.0	0.00	0.00	0.01	0.17	0.99	0.30	3.57 0	3.91
DAX - VDAX	DQhit	0.10 0.03 0.91	0.36 0.03 0.91 0	0.02 0.03 0.	.85 0.20	0.11 0.88	0.07 0.11	0.86	0.07 0.1	2 0.88	0.12	0.50	0.95 0.	08 0.54	0.95	0.15	0.75	0.77	0.15 0.7	5 0.80	0.17	0.06	0.75	0.06	0.79 (0.74 0.4	2 0.70	0.75	0.39	0.22	0.92	0.68	0.38 0	J.91
	DQVaR	0.24 0.06 0.99	0.61 0.06 0.99 0	0.07 0.06 0.	.98 0.44	0.20 0.99	0.10 0.19	0.98	0.15 0.2	0 0.99	0.01	0.70	0.81 0.	00 0.64	0.81	0.33	0.63	0.92	0.36 0.6	4 0.94	0.00	0.11	0.92	0.17	0.64 (0.91 0.0	i9 0.62	0.92	0.59	0.31	0.99	0.00	0.51 0	0.99
	MCuc	0.39 0.75 0.67	0.72 0.74 0.73	0.33 0.74 0.	07 0.26	0.76 0.38	0.02 0.69	0.12	0.06 0.8	2 0.35	0.01	0.02	0.43 0.	00 0.00	0.24	0.95	0.20	0.00	0.89 0.1	7 0.00	0.11	0.27	0.00	0.80	0.11 0	.00 0.1	5 0.29	0.00	0.68	0.14	0.79	0.19	0.28 (0.74
	MCind	0.36 0.43 0.23	0.65 0.77 0.22	0.32 0.71 0	58 0.22	0.11 0.17	0.08 0.38	0.53	0.16 0.7	2 0.60	0.20	0.59	0.99 0	35 0.19	0.00	0.01	0.88	0.49	0.01 0.8	7 0.64	0.04	0.52	0.43	0.00	0.78 (36 0.0	0 0.87	0.49	0.00	0.94	0.11	0.01	0.86 (0.07
	MC.	0.50 0.10 0.20	0.50 0.45 0.45	0.02 0.71 0.	01 0.42	0.00 0.22	0.17 0.70	0.00	0.20 0.7	5 0.70	0.41	0.00	0.00 0.	00 0.10	0.01	0.01	0.00	0.00	0.01 0.0	0.01	0.07	0.05	0.00	0.01	0.40 0	70 01	0 0.00	0.00	0.00	0.10	0.00	0.01	0.00 (0.14
	MUCC	0.74 0.80 0.46	0.70 0.45 0.45 0	0.65 0.59 0.	.81 0.45	0.20 0.33	0.17 0.79	0.92	0.32 0.3	5 0.78	0.41	0.84	0.02 0.	07 0.38	0.01	0.01	0.24	0.98	0.01 0.2	5 0.71	0.07	0.95	0.88	0.01	0.40 (.70 0.0	0.20	0.99	0.00	0.12	0.22	0.01	J.28 U	J.14
	Exc.	180 39 5	180 37 5	178 37	5 187	37 5	201 41	7	205 36	i 6	153	25	7 1	57 29	6	161	34	10	162 33	5 11	153	28	8	166	38	12 10	3 36	11	164	36	12	156	27	7
	LRuc	0.76 0.53 0.46	0.76 0.76 0.46	0.88 0.76 0.	.46 0.40	0.76 0.46	0.06 0.34	0.10	0.03 0.9	0 0.23	0.07	0.07	0.10 0.	13 0.28	0.23	0.26	0.86	0.00	0.30 0.9	9 0.00	0.08	0.21	0.04	0.47	0.62 0	.00 0.3	4 0.87	0.00	0.38	0.87	0.00	0.13	0.15 0	0.10
	LRind	0.21 0.45 0.91	0.21 0.41 0.91 0	0.03 0.41 0.	91 0.11	0.41 0.91	0.10 0.09	0.87	0.23 0.3	8 0.89	0.10	0.55	0.87 0	08 0.49	0.89	0.82	0.41	0.81	0.57 0.4	0 0.79	0.60	0.22	0.85	0.68	0.36 (0.77 0.1	7 0.39	0.79	0.62	0.39	0.77	0.68	0.21 (0.87
	LRcc	0.44 0.62 0.75	0.44 0.68 0.75	0.08 0.68 0	75 0.20	0.68 0.75	0.05 0.16	0.26	0.05 0.6	8 0.48	0.05	0.16	0.26 0	07 0.43	0.48	0.52	0.71	0.02	0.50 0.7	0 0.01	0.18	0.22	0.12	0.70	0.58 0	00 0.	2 0.68	0.01	0.60	0.68	0.00	0.29	0.16 0	0.26
DH VYD	DOLA	0.00 0.46 0.01	0.00 0.41 0.01 4	0.00 0.00 0.	01 0.10	0.41 0.01	0.11 0.10	0.20	0.00 0.0	0 0.10	0.11	0.75	0.20 0.	00 0.40	0.00	0.02	0.40	0.01	0.50 0.1	0 0.70	0.01	0.02	0.07	0.00	0.00 0	77 0.	0.00	0.70	0.00	0.00	0.77	0.00	0.01 (0.07
DJI - VAD	DQIII	0.22 0.40 0.91	0.22 0.41 0.91 0	0.03 0.41 0.	.91 0.12	0.41 0.91	0.11 0.10	0.87	0.24 0.3	9 0.89	0.11	0.00	0.87 0.	09 0.49	0.89	0.82	0.42	0.01	0.38 0.4	0 0.19	0.01	0.25	0.80	0.08	0.50 0	0.0	0.59	0.19	0.05	0.39	0.11	0.09	3.21 0	3.01
	DQVaR	0.21 0.66 0.03	0.39 0.64 0.03 0	0.06 0.63 0.	03 0.20	0.62 0.03	0.10 0.21	0.03	0.24 0.5	9 0.01	0.22	0.01	0.03 0.	22 0.05	0.24	0.81	0.56	0.97	0.79 0.5	5 0.96	0.13	0.41	0.92	0.84	0.51 (0.96 0.8	9 0.54	0.96	0.83	0.53	0.96	0.05	J.39 0	J.00
	MCuc	0.73 0.51 0.33	0.73 0.79 0.33 0	0.89 0.78 0.	.47 0.38	0.73 0.51	0.06 0.36	0.11	0.03 0.8	3 0.19	0.08	0.08	0.09 0.	15 0.30	0.18	0.25	0.87	0.00	0.31 0.9	7 0.00	0.09	0.23	0.06	0.47	0.60 0	.00 0.3	3 0.87	0.00	0.37	0.78	0.00	0.13	0.16 0	0.10
	MCind	0.01 0.00 0.32	0.01 0.00 0.33 0	0.01 0.08 0.	.34 0.01 (0.00 0.62	0.01 0.00	0.15	0.01 0.0	0 0.26	0.58	0.01	0.70 0.	73 0.11	0.87	0.58	0.38	0.02	0.62 0.4	4 0.01	0.30	0.51	0.04	0.67	0.06 0	.03 0.1	7 0.24	0.01	0.14	0.30	0.02	0.01	0.06 0	0.11
	MCcc	0.02 0.00 0.67	0.02 0.00 0.66 0	01 017 0	66 0.02 0	0 00 0 77	0.02 0.00	0.31	0.02 0.0	0 0.52	0.85	0.02	0.61 0	53 0.25	0.26	0.85	0.76	0.04	0.78 0.9	3 0.02	0.62	0.97	0.09	0.66	0.13 (05 0 3	4 0.50	0.02	0.27	0.59	0.05	0.01	0.13 f	0.22
	Eno	180 20 6	172 22 5	179 21	7 171	20 6	107 20	7	102 010	7	140	20	6 1	10 97	6	171	20	10	179 29	0 0102	160	21	2	177	20	7 19	0 20	0.02	179	20	7	164	700 0	6
	EXC.	180 30 0	113 33 3	172 31	1 1/1	39 0	197 30		195 20		140	22	0 1	19 21	0	1/1	- 39	10	1/6 30		109	31		111	- 30	/ 10	0 30	9	110			104	20	
	LRuc	0.53 0.44 0.21	0.93 0.81 0.43 0	0.99 0.56 0.	.09 0.95	0.44 0.21	0.05 0.44	0.09	0.10 0.2	6 0.09	0.01	0.02	0.21 0.	07 0.19	0.21	0.97	0.43	0.00	0.61 0.5	3 0.09	0.85	0.57	0.81	0.67	0.53 (1.09 0.3	0.53	0.01	0.61	0.53	0.09	0.56	J.26 0	J.21
	LRind	0.14 0.27 0.01	0.43 0.33 0.00	0.25 0.29 0.	01 0.23	0.08 0.01	0.16 0.27	0.01	0.20 0.2	3 0.01	0.43	0.13	0.01 0.	28 0.21	0.01	0.87	0.08	0.02	0.55 0.4	4 0.01	0.21	0.03	0.00	0.53	0.44 0	.01 0.0	i0 0.44	0.02	0.55	0.44	0.01	0.26	0.23 0	J.01
	LRcc	0.28 0.40 0.01	0.73 0.60 0.01	0.51 0.48 0.	01 0.49	0.16 0.01	0.06 0.40	0.01	0.12 0.2	6 0.01	0.03	0.02	0.01 0.	10 0.19	0.01	0.99	0.16	0.00	0.74 0.6	1 0.01	0.44	0.08	0.00	0.75	0.61 0	.01 0.1	0 0.61	0.00	0.74	0.61	0.01	0.45	0.26 0	0.01
Nikkei - VXJ	DOhit	0.15 0.27 0.01	0.44 0.33 0.00 (0.26 0.29 0.	01 0.25	0.08 0.01	0.17 0.27	0.01	0.22 0.2	3 0.01	0.45	0.13	0.01 0.	29 0.21	0.01	0.87	0.08	0.02	0.56 0.4	4 0.01	0.22	0.03	0.00	0.54	0.44 0	.01 0.0	0.44	0.02	0.56	0.44	0.01	0.27	0.23 €	0.01
	DOVeP	0.21 0.46 0.02	0.45 0.52 0.02	0.20 0.40 0	02 0.51	0.18 0.02	0.20 0.42	0.02	0.26 0.2	0 0.02	0.12	0.00	0.00 0	10 0.01	0.00	0.90	0.16	0.07	0.55 0.5	6 0.02	0.06	0.08	0.00	0.20	0.57 0	02 0	1 0.55	0.06	0.20	0.57	0.02	0.05	0.40 C	0.02
	DQVan	0.31 0.40 0.02	0.43 0.33 0.02	0.59 0.49 0.	03 0.31	0.18 0.02	0.29 0.45	0.03	0.20 0.3	9 0.03	0.15	0.00	0.00 0.	10 0.01	0.00	0.29	0.10	0.07	0.00 0.0	0.03	0.00	0.08	0.00	0.59	0.07 0	.03 0.4	1 0.00	0.00	0.59	0.07	0.03	0.03	J.40 U	5.02
	MCuc	0.54 0.46 0.18	0.96 0.79 0.49 0	0.99 0.58 0.	.08 0.96	0.45 0.25	0.06 0.42	0.08	0.10 0.2	8 0.11	0.01	0.02	0.17 0.	07 0.21	0.19	0.97	0.38	0.00	0.58 0.5	1 0.08	0.82	0.53	0.97	0.64	0.50 (0.05 0.4	6 0.48	0.01	0.60	0.46	0.07	0.59	J.25 0	0.21
	MCind	0.42 0.78 0.81	0.46 0.50 0.42 0	0.31 0.42 0.	.71 0.20	0.50 0.53	0.10 0.45	0.75	0.05 0.5	0 0.75	0.56	0.38	0.02 0.	42 0.40	0.02	0.01	0.87	0.76	0.01 0.9	1 0.67	0.00	0.07	0.04	0.02	0.91 (0.67 0.0	0 0.88	0.69	0.02	0.91	0.67	0.01	0.08 0	0.25
	MCcc	0.85 0.45 0.38	0.92 1.00 0.84	0.61 0.82 0.	.59 0.40	0.98 0.94	0.22 0.87	0.50	0.09 0.9	9 0.49	0.87	0.78	0.04 0.	82 0.79	0.04	0.03	0.25	0.48	0.01 0.1	9 0.68	0.01	0.14	0.08	0.04	0.17 (.68 0.0	0 0.23	0.63	0.04	0.18	0.68	0.01	0.17 0	0.50
	Exc	137 26 2	141 27 2	127 28	3 133	25 3	157 29	5	157 25	4	140	22	6 1	21 17	2	121	26	5	123 29	5	110	29	3	122	27 5	00 15	1 27	5	123	28	5	110	23	2
	I Pase	0.00 0.78 0.64	0.74 0.02 0.64 4	0.27 0.01 0	ee 0.71	0.62 0.99	0.00 0.77	0.99	0.00 0.6	2 0.49	0.01	0.02	0.21 0	15 0.02	0.64	0.17	0.90	0.99	0.22 0.5	0 0 99	0.02	0.74	0.97	0.90	0.06 (199 0	7 0.06	0.99	0.92	0.90	0.99	0.02	0.40 (0.64
	LINUC	0.55 0.78 0.04	0.74 0.93 0.04 0	0.57 0.91 0.	.00 0.71	0.05 0.88	0.09 0.11	0.22	0.09 0.0	5 0.46	0.01	0.02	0.21 0.	10 0.03	0.04	0.17	0.00	0.22	0.25 0.6	9 0.22	0.02	0.74	0.87	0.20	0.90 0	.22 0.	1 0.90	0.22	0.25	0.89	0.22	0.02	J.40 0	3.04
	LRind	0.22 0.48 0.96	0.17 0.46 0.96 0	0.64 0.45 0.	.94 0.03	0.50 0.94	0.47 0.43	0.89	0.72 0.5	0 0.91	0.43	0.13	0.01 0.	26 0.65	0.96	0.52	0.25	0.89	0.47 0.2	9 0.89	0.45	0.32	0.94	0.84	0.27 (0.89 0.8	2 0.27	0.89	0.47	0.29	0.89	0.83	J.19 0	J.96
	LRcc	0.47 0.75 0.89	0.37 0.76 0.89	0.59 0.74 0.	.99 0.10	0.71 0.99	0.18 0.70	0.47	0.22 0.7	1 0.77	0.03	0.02	0.01 0.	18 0.09	0.89	0.31	0.50	0.46	0.38 0.5	7 0.46	0.04	0.57	0.98	0.43	0.54 (0.46 0.3	1 0.54	0.46	0.38	0.57	0.46	0.05	0.29 0	0.90
Nasdaq - VXN	DQhit	0.23 0.48 0.96	0.18 0.47 0.96	0.65 0.45 0.	.94 0.04	0.50 0.94	0.48 0.43	0.89	0.73 0.5	0 0.91	0.45	0.13	0.01 0.	26 0.65	0.96	0.53	0.25	0.89	0.48 0.3	0 0.89	0.46	0.32	0.94	0.84	0.27 (.89 0.3	3 0.27	0.89	0.48	0.30	0.89	0.83	0.19 6	0.96
	DOVaR	0.27 0.03 0.98	0.21 0.03 0.98	0.36 0.00 0	76 0.03	0.10 0.56	0.59 0.03	0.93	0.55 0.0	9 0.98	0.13	0.00	0.00 0.	53 0.90	0.95	0.59	0.39	0.99	0.60 0.4	4 0.99	0.22	0.52	1.00	0.88	0.42 0	.99 0.0	5 0.41	0.99	0.60	0.44	0.99	0.14	0.35 1	1.00
	MCue	0.07 0.92 0.95	0.74 0.07 0.77	0.27 0.08 0	95 0.71	0.66 0.08	0.00 0.76	0.92	0.08 0.7	0 0.42	0.01	0.02	0.17 0	15 0.04	0.91	0.16	0.79	0.99	0.99 0.9	4 0.17	0.02	0.79	0.50	0.90	0.04	194 0	6 0.00	0.19	0.94	0.90	0.14	0.02	0.42 (0.72
	MCdC	0.97 0.83 0.85	0.14 0.97 0.77 0	0.57 0.98 0.	.00 0.71	0.00 0.98	0.09 0.76	0.23	0.06 0.7	0 0.45	0.01	0.02	0.17 0.	10 0.04	0.81	0.10	0.18	0.42	0.22 0.8	- 0.17	0.02	0.78	0.09	0.20	0.94 (1.24 0.	0 0.92	0.18	0.24	0.00	0.14	0.02	0.40 0	5.13
	MCind	0.66 0.40 0.70	0.94 0.33 0.70 0	0.52 0.24 0.	.46 0.02 0	0.03 0.42	0.34 0.00	0.18	0.21 0.0	0.18	0.57	0.39	0.02 0.	12 0.06	0.69	0.00	0.00	0.16	0.00 0.0	0.17	0.00	0.03	0.40	0.00	0.00 (0.0	0.01	0.17	0.00	0.00	0.16	0.00	J.07 0	J.68
	MCcc	0.66 0.80 0.62	0.12 0.65 0.62	0.95 0.47 0.	.92 0.05	0.06 0.82	0.71 0.00	0.35	0.42 0.0	1 0.36	0.86	0.77	0.04 0.	25 0.11	0.63	0.00	0.00	0.34	0.00 0.0	0 0.34	0.00	0.06	0.77	0.00	0.01 (.35 0. 0	0 0.01	0.34	0.00	0.00	0.34	0.00	0.15 0	J.63
																														_	-		-	

Table 5: Results (in the form of p-values) for in-sample VaR accuracy for all GARCH, BEGE, univariate and bivariate Hawkes-POT models. Column headings denote the model at each level of significance α , the rows in each panel denote the test. The number of exceptions observed (Exc.) are also reported above the test results. All tests with a p-value less than 5% are shown in bold to highlight where the rejections of the accuracy tests are occurring.

Returns	Statistics										1	/olatilit;	y Model	ls (Out-	of-sam	ple)											U:	nivariat	e Hawk	es-POT	(Out-of	-sample)					Bivari	late Hav	wkes-P(JT (Out	i-of-san	nple)			
rectarins	Statistics	G.	RGAR	CH	GJR	GARCI	H+IV	1	EGARC	H	EG	ARCH	+IV		GARCI	H	GA	ARCH+	IV		BEGE		E	EGE+I	V		M1			M2			13		M1			M2			M3			M4	
-	α ₋ level	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95	0.00	0.000	0.95 0	00 D 00	9 0.95	0.00	0.000	0.95	0.00	0.000	0.95 (1 00 0	1 999	0.95 (1 99 C	0.000
	n- level	0.30	0.33	0.333	0.30	0.33	0.333	0.30	0.33	0.333	0.30	0.33	0.333	0.30	0.33	0.333	0.30	0.33	0.000	0.30	0.00	0.000	0.50	0.33	0.333	0.30	0.00	0.333	0.30	0.00	0.333	10.00 0	33 0.32	0.00	0.33	0.000	0.30	0.33	0.333	0.30 0	1.00 0	1.000	0.00 0		3.333
	Exc.	Э	1	0	5	0	0	5	0	0	10	0	0	3	0	0	3	0	0	19	10	4	18	2	0	- 38	8	1	41	17	2	40	1 1	30	- 7	1	30	7	1	23	3	1	23	5	1
	LRuc	0.00	0.03	0.32	0.00	0.00	0.32	0.00	0.00	0.32	0.00	0.00	0.32	0.00	0.00	0.32	0.00	0.00	0.32	0.19	0.05	0.00	0.13	0.12	0.32	0.02	0.23	0.54	0.00	0.00	0.11	0.01 0	02 0.5	4 0.35	0.41	0.54	0.35	0.41	0.54	0.63 0	J.32 (0.54	0.63 0	J.98 (0.54
	LRind	0.75	0.95	1.00	0.75	1.00	1.00	0.75	1.00	1.00	0.52	1.00	1.00	0.85	1.00	1.00	0.85	1.00	1.00	0.75	0.52	0.80	0.25	0.90	1.00	0.56	0.61	0.95	0.40	0.28	0.90	0.92 0	48 0.9	5 0.86	0.66	0.95	0.86	0.66	0.95	0.14 (0.85	0.95	0.14 f) 75 /	0.95
	I.D	0.00	0.00	0.01	0.00	0.01	0.61	0.00	0.01	0.01	0.00	0.01	0.01	0.00	0.01	0.01	0.00	0.01	0.61	0.41	0.10	0.01	0.10	0.20	0.01	0.07	0.40	0.02	0.01	0.00	0.00	0.00 0	00 0.0	0.04	0.07	0.02	0.04	0.05	0.02	0.20 (0.00	0.02	0.20 (0.05	0.02
	LACC	0.00	0.09	0.01	0.00	0.01	0.01	0.00	0.01	0.01	0.00	0.01	0.01	0.00	0.01	0.01	0.00	0.01	0.01	0.41	0.12	0.01	0.10	0.30	0.01	0.05	0.42	0.85	0.01	0.00	0.28	0.02 0	00 0.8	0.04	0.05	0.85	0.04	0.65	0.85	0.50 0	J.00 I	0.85	0.50 0	7.99 (0.85
S&P500 - VIX	DQhit	0.76	0.95	1.00	0.76	1.00	1.00	0.76	1.00	1.00	0.53	1.00	1.00	0.85	1.00	1.00	0.85	1.00	1.00	0.75	0.53	0.80	0.26	0.90	1.00	0.58	0.61	0.95	0.41	0.28	0.90	0.92 0	49 0.9	5 0.87	0.66	0.95	0.87	0.66	0.95	0.15 0	J.85 /	0.95	0.15 ().75 (0.95
	DQVaR	0.25	0.56	1.00	0.79	1.00	1.00	0.30	1.00	1.00	0.42	1.00	1.00	0.88	1.00	1.00	0.87	1.00	1.00	1.00	1.00	1.00	0.17	0.91	1.00	0.84	0.71	0.99	0.56	0.35	0.98	0.98 0	59 0.9	3 0.90	0.79	1.00	0.90	0.80	1.00	0.30 0	J.95	1.00	0.32 (J.87 (0.99
	MCue	0.00	0.05	0.75	0.00	0.01	0.87	0.00	0.00	0.99	0.00	0.00	0.25	0.00	0.00	0.40	0.00	0.00	0.12	0.19	0.04	0.00	0.15	0.16	1.00	0.01	0.25	0.63	0.00	0.00	0.09	0 00 0	02 0.4	3 0.33	0.32	0.38	0.37	0.29	0.68	0.69 f	n 39	0.60	0.72 (a 91 – 7	0.32
	MOUL	0.50	0.00	1.00	0.50	1.00	1.00	0.50	1.00	1.00	0.00	1.00	1.00	0.00	1.00	1.00	0.00	1.00	1.00	0.15	0.55	0.00	0.10	0.01	1.00	0.001	0.20	0.00	0.10	0.00	0.00	0.00 0	02 0.1	0.00	0.02	0.00	0.01	0.20	0.00	0.00 0	0.40	0.00	0.00 /		0.02
	MCind	0.52	0.36	1.00	0.78	1.00	1.00	0.52	1.00	1.00	0.75	1.00	1.00	0.97	1.00	1.00	0.97	1.00	1.00	0.45	0.57	0.88	0.94	0.91	1.00	0.23	0.76	0.91	0.13	0.97	0.68	0.08 0	95 0.9	0.42	0.87	0.91	0.42	0.88	0.90	0.84 0	J.42 (0.91	0.69 0	1.11 (0.91
	MCcc	0.97	0.72	1.00	0.45	0.00	0.88	0.96	0.00	1.00	0.53	1.00	0.41	0.06	0.00	0.61	0.06	0.00	1.00	0.88	0.83	0.23	0.12	0.18	1.00	0.45	0.48	0.20	0.26	0.06	0.67	0.16 0	10 0.2	0.87	0.24	0.18	0.87	0.24	0.18	0.34 (J.83 (0.19	0.61 ().24 (0.19
	Exc.	18	5	0	19	6	0	20	5	0	24	6	0	21	5	0	22	5	0	18	2	0	21	5	0	27	3	0	29	2	0	29	2 0	30	3	0	29	2	0	29	2	0	22	2	0
	I Duo	0.12	0.00	0.29	0.10	0.67	0.29	0.98	0.00	0.29	0.89	0.67	0.29	0.20	0.00	0.29	0.59	0.00	0.29	0.11	0.19	0.21	0.26	0.08	0.21	0.60	0.22	0.29	0.42	0.19	0.29	0.42 0	19 0.2	0.20	0.22	0.29	0.42	0.19	0.20	0.42 (0.19	0.29	0.59 (a 19 - 7	0.29
	LIQUE	0.15	0.99	0.32	0.19	0.07	0.32	0.28	0.99	0.32	0.82	0.07	0.32	0.59	0.99	0.32	0.32	0.99	0.32	0.11	0.12	0.51	0.50	0.98	0.51	0.09	0.55	0.52	0.45	0.12	0.32	0.45 0	12 0.5	2 0.32	0.55	0.32	0.45	0.12	0.32	0.45 0).12 (0.52	0.02 0	.12 (0.52
	LRind	0.25	0.75	1.00	0.75	0.70	1.00	0.82	0.75	1.00	0.88	0.70	1.00	0.18	0.75	1.00	0.16	0.75	1.00	0.26	0.90	1.00	0.89	0.75	1.00	0.67	0.85	1.00	0.55	0.90	1.00	0.55 0	90 1.0	0.49	0.85	1.00	0.55	0.90	1.00	0.55 0	0.90	1.00	0.97 0	J.90 I	1.00
	LRcc	0.16	0.95	0.61	0.41	0.85	0.61	0.54	0.95	0.61	0.96	0.85	0.61	0.27	0.95	0.61	0.30	0.95	0.61	0.15	0.29	0.60	0.65	0.95	0.60	0.85	0.61	0.61	0.61	0.30	0.61	0.61 0	30 0.6	0.49	0.61	0.61	0.61	0.30	0.61	0.61 0	J.30 (0.61	0.82 0).30 (0.61
DAX - VDAX	DOhit	0.26	0.75	1.00	0.75	0.70	1.00	0.82	0.75	1.00	0.89	0.70	1.00	0.19	0.75	1.00	0.17	0.75	1.00	0.26	0.90	1.00	0.89	0.75	1.00	0.68	0.85	1.00	0.56	0.90	1.00	0.56 0	90 1.0	0.51	0.85	1.00	0.56	0.90	1.00	0.56 (0.90	1.00	0.97 (0.90	1.00
	DOVeP	0.41	0.02	1.00	0.59	0.02	1.00	0.76	0.01	1.00	0.01	0.17	1.00	0.20	0.09	1.00	0.26	0.02	1.00	0.59	0.09	1.00	0.74	0.12	1.00	0.17	0.05	1.00	0.97	0.07	1.00	0.97 0	07 1.0	0.99	0.01	1.00	0.96	0.07	1.00	0.97 (0.07	1.00	0.90 (1.07	1.00
	DQvan	0.41	0.03	1.00	0.08	0.02	1.00	0.70	0.01	1.00	0.91	0.17	1.00	0.59	0.02	1.00	0.30	0.03	1.00	0.02	0.98	1.00	0.74	0.15	1.00	0.17	0.90	1.00	0.27	0.97	1.00	0.27 0	97 1.0	0.22	0.91	1.00	0.20	0.97	1.00	0.27 0	5.91	1.00	0.20 0	1.91 .	1.00
	MCuc	0.16	0.90	1.00	0.22	0.66	1.00	0.26	0.89	1.00	0.91	0.74	1.00	0.45	0.94	1.00	0.56	0.98	1.00	0.13	0.17	1.00	0.34	0.87	1.00	0.63	0.43	1.00	0.37	0.10	1.00	0.46 0	22 1.0	0.27	0.35	1.00	0.44	0.12	1.00	0.45 0).17	1.00	0.60 0).16	1.00
	MCind	0.94	0.40	1.00	0.89	0.27	1.00	0.86	0.42	1.00	0.60	0.28	1.00	0.83	0.40	1.00	0.75	0.40	1.00	0.43	0.85	1.00	0.81	0.92	1.00	0.40	0.41	1.00	0.31	0.60	1.00	0.31 0	59 1.0	0.25	0.43	1.00	0.32	0.61	1.00	0.30 0	J.60	1.00	0.79 0	J.06 '	1.00
	MCcc	0.11	0.79	1.00	0.20	0.53	1.00	0.29	0.87	1.00	0.79	0.54	1.00	0.34	0.80	1.00	0.48	0.81	1.00	0.87	0.31	1.00	0.37	0.16	1.00	0.78	0.82	1.00	0.60	0.81	1.00	0.63 0	80 1.0	0.50	0.84	1.00	0.60	0.80	1.00	0.61 (0.80	1.00	0.42 (a.11 - 1	1.00
	Eno	15	1	0	15	1	0	14	0.01	0	10	1	1	10	9	0	17	0	0	14	1	0	14	1	0	94	2	0	94	2	0	10	2 0	95	2	0	20	2	0	- 25	2	0	10	0	0
	EXC.	10	1	0	10	1		14			10	1	1	19			17	0	0	14	1		14	1	0	24			24	3	0	19	2 0	20		0	20		0	20			19		
	LRuc	0.03	0.03	0.32	0.03	0.03	0.32	0.01	0.12	0.32	0.13	0.03	0.54	0.19	0.12	0.32	0.08	0.00	0.32	0.01	0.03	0.32	0.01	0.03	0.32	0.84	0.33	0.32	0.84	0.33	0.32	0.20 0	13 0.3	2 0.99	0.33	0.32	0.99	0.33	0.32	0.99 0	0.33 (0.32	0.20 0).13 (0.32
	LRind	0.34	0.95	1.00	0.34	0.95	1.00	0.37	0.90	1.00	0.67	0.95	0.95	0.22	0.90	1.00	0.27	1.00	1.00	0.37	0.95	1.00	0.37	0.95	1.00	0.88	0.85	1.00	0.88	0.85	1.00	0.75 0	90 1.0	0.52	0.85	1.00	0.52	0.85	1.00	0.52 (J.85 1	1.00	0.75 0	J.90 2	1.00
	LRcc	0.05	0.09	0.61	0.05	0.09	0.61	0.03	0.30	0.61	0.28	0.09	0.82	0.20	0.30	0.61	0.12	0.01	0.61	0.03	0.09	0.61	0.03	0.09	0.61	0.97	0.62	0.61	0.97	0.62	0.61	0.42 0	31 0.6	0.81	0.62	0.61	0.81	0.62	0.61	0.81 (0.62	0.61	0.42 (J.31 (0.61
DIL - VXD	DOhit	0.35	0.95	1.00	0.35	0.95	1.00	0.38	0.90	1.00	0.68	0.95	0.95	0.23	0.90	1.00	0.29	1.00	1.00	0.38	0.95	1.00	0.38	0.95	1.00	0.88	0.85	1.00	0.88	0.85	1.00	0.75 0	90 1.0	0.53	0.85	1.00	0.53	0.85	1.00	0.53 (0.85	1.00	0.75 (1.90	1.00
	DOVeP	0.45	0.09	1.00	0.51	0.06	1.00	0.50	0.00	1.00	0.79	0.07	0.02	0.04	0.02	1.00	0.00	1.00	1.00	0.69	0.99	1.00	0.69	0.94	1.00	0.22	0.06	1.00	0.20	0.07	1.00	0.90 0	00 10	0.99	0.07	1.00	0.21	0.07	1.00	0.29 (0.07	1.00	0.17 (1.00	1.00
	DQvan	0.40	0.08	1.00	0.01	0.00	1.00	0.00	0.00	1.00	0.78	0.07	0.08	0.04	0.02	1.00	0.09	1.00	1.00	0.08	0.22	1.00	0.08	0.24	1.00	0.55	0.90	1.00	0.59	0.97	1.00	0.20 0	96 1.0	0.28	0.97	1.00	0.51	0.97	1.00	0.32 0	5.91	1.00	0.17 0	1.96	1.00
	MCuc	0.02	0.03	1.00	0.04	0.06	1.00	0.02	0.17	1.00	0.16	0.04	0.57	0.23	0.21	1.00	0.09	1.00	1.00	0.02	0.03	1.00	0.01	0.04	1.00	0.87	0.45	1.00	0.80	0.34	1.00	0.18 0	08 1.0	0.99	0.38	1.00	0.96	0.44	1.00	0.96 0).54	1.00	0.18 0).17	1.00
	MCind	0.74	0.35	1.00	0.74	0.36	1.00	0.80	0.61	1.00	0.78	0.35	0.36	0.72	0.61	1.00	0.56	0.00	1.00	0.80	0.37	1.00	0.80	0.36	1.00	0.68	0.41	1.00	0.68	0.42	1.00	0.40 0	59 1.0	0.59	0.41	1.00	0.59	0.42	1.00	0.59 0	J.41	1.00	0.39 0	J.58 [*]	1.00
	MCcc	0.53	0.71	1.00	0.53	0.72	1.00	0.40	0.78	1.00	0.44	0.72	0.72	0.56	0.78	1.00	0.87	1.00	1.00	0.43	0.73	1.00	0.40	0.72	1.00	0.62	0.84	1.00	0.63	0.83	1.00	0.79 0	82 1.0	0.83	0.83	1.00	0.82	0.84	1.00	0.84 (0.82	1.00	0.80 ().82	1.00
	Exe	10	1	1	10	1	1	18	1	1	25	2	1	20	1	1	10	1	1	18	2	1	18	2	1	28	6	1	97	6	1	97	7 1	26	6	- 1	- 27	7	1	26	6	1	31	6	1
	LAC.	0.10	0.00	0.54	0.10	0.00	0.54	0.12	0.00	0.54	0.00	0.10	0.54	0.00	0.02	0.74	0.10	0.00	0.54	0.15	0.12	0.52	0.15	0.12	0.52	0.50	0.64	0.52	0.02	0.64	0 72	0.02 0	20 07	20	0.64	0.52	0.02	0.20	0.52	0.70 (0.64	0.52	0.01 (0.04	0.52
	LIQUE	0.19	0.03	0.54	0.19	0.03	0.54	0.15	0.03	0.54	0.98	0.12	0.54	0.28	0.03	0.54	0.19	0.03	0.54	0.15	0.15	0.55	0.15	0.15	0.55	0.50	0.04	0.55	0.65	0.04	0.55	0.65 0	38 0.5	0.78	0.04	0.55	0.05	0.58	0.55	0.78 0	J.04 I	0.55	0.21 0	7.04 U	0.55
	LRind	0.71	0.95	0.95	0.71	0.95	0.95	0.63	0.95	0.95	0.85	0.90	0.95	0.78	0.95	0.95	0.71	0.95	0.95	0.68	0.90	0.95	0.68	0.90	0.95	0.63	0.01	0.95	0.69	0.01	0.95	0.69 0	01 0.9	5 0.76	0.05	0.95	0.69	0.08	0.95	0.76 (J.05 (0.95	0.45 0).70 (0.95
	LRcc	0.40	0.09	0.82	0.40	0.09	0.82	0.28	0.09	0.82	0.98	0.30	0.82	0.54	0.09	0.82	0.40	0.09	0.82	0.33	0.32	0.82	0.33	0.32	0.82	0.71	0.01	0.82	0.82	0.01	0.82	0.82 0	01 0.8	2 0.92	0.14	0.82	0.82	0.15	0.82	0.92 0	0.14 (0.82	0.34 (J.83 (0.82
Nikkei - VXI	DOhit	0.71	0.95	0.95	0.71	0.95	0.95	0.64	0.95	0.95	0.85	0.90	0.95	0.79	0.95	0.95	0.71	0.95	0.95	0.69	0.90	0.95	0.69	0.90	0.95	0.64	0.01	0.95	0.70	0.01	0.95	0.70 0	01 0.9	5 0.77	0.06	0.95	0.70	0.08	0.95	0.77 (a ne 👘	0.95	0.47 (1.70 /	0.95
	DOVeP	0.02	0.71	0.60	0.01	0.75	0.76	0.00	0.66	0.66	0.07	0.20	0.00	0.05	0.06	0.06	0.02	0.06	0.06	0.92	0.59	0.24	0.92	0.59	0.24	0.95	0.02	1.00	0.00	0.02	1.00	0.96 0	02 1.0	0.04	0.15	1.00	0.00	0.91	1.00	0.04 (0.15	1.00	0.72 (197	1.00
	DQvan	0.95	0.71	0.09	0.91	0.75	0.70	0.88	0.00	0.00	0.97	0.30	0.99	0.90	0.90	0.90	0.95	0.90	0.90	0.85	0.32	0.54	0.65	0.02	0.34	0.80	0.02	1.00	0.90	0.02	1.00	0.80 0	03 1.0	0.94	0.13	1.00	0.00	0.21	1.00	0.94 0	J.10 .	1.00	0.15 0	401 -	1.00
	MCuc	0.22	0.06	0.49	0.24	0.07	0.64	0.14	0.02	0.58	0.95	0.22	0.28	0.35	0.07	0.68	0.22	0.07	0.60	0.14	0.09	0.72	0.18	0.20	0.66	0.54	0.63	0.22	0.67	0.63	0.66	0.68 0	30 0.6	5 0.77	0.61	0.51	0.62	0.28	0.41	0.72 0	J.65 F	0.46	0.17 0	1.53 (0.27
	MCind	0.28	0.30	0.29	0.33	0.30	0.30	0.29	0.30	0.30	0.32	0.25	0.30	0.55	0.30	0.30	0.66	0.31	0.31	0.46	0.26	0.31	0.46	0.26	0.31	0.22	0.25	0.30	0.35	0.25	0.31	0.25 0	25 0.3	0.34	0.10	0.31	0.38	0.51	0.30	0.33 0	J.10 (0.29	0.17 0).17 (0.30
	MCcc	0.54	0.60	0.59	0.67	0.59	0.60	0.58	0.59	0.60	0.64	0.51	0.61	0.89	0.59	0.61	0.69	0.59	0.61	0.92	0.51	0.60	0.91	0.52	0.60	0.43	0.50	0.62	0.71	0.50	0.63	0.52 0	51 0.6	2 0.65	0.20	0.60	0.73	0.97	0.61	0.66 (0.20	0.61	0.31 ().34 /	0.61
	Exe	28	9	1	28	9	0.00	34	11	4	31	16	2	31	11	2	31	11	2	13.00	1.00	0.00	18.00	2.00	0.00	25	3	0	26	5	0	22	2 0	26	5	0	26	5	0	26	5	0	24	6	0
	LAC.	0.50	0.11	0.54	0.50	0.11	0.20	0.00	0.00	0.00	0.04	0.00	0.11	0.04	0.00	0.11	0.04	0.00	0 11	0.01	0.00	0.00	0.12	0.10	0.00	0.00	0.20	0.20	0.72	0.07	0.20	0.00 0	14 0.2	0.72	0.05	0.20	0.72	0.05	0.20	0.72 (0.05	0.20	0.04 (0.00	0.20
	LIQUE	0.50	0.11	0.54	0.50	0.11	0.52	0.08	0.02	0.00	0.24	0.00	0.11	0.24	0.02	0.11	0.24	0.02	0.11	0.01	0.03	0.52	0.15	0.12	0.32	0.89	0.30	0.32	0.75	0.95	0.32	0.62 0	14 0.5	2 0.75	0.95	0.32	0.75	0.95	0.52	0.75 0	J.95 I	0.32	0.94 0	7.02 U	0.32
	LRind	0.07	0.57	0.95	0.07	0.57	1.00	0.30	0.48	0.80	0.04	0.30	0.90	0.44	0.48	0.90	0.44	0.48	0.90	0.34	0.95	1.00	0.67	0.90	1.00	0.82	0.85	1.00	0.75	0.75	1.00	0.96 0	90 1.0	0.75	0.75	1.00	0.75	0.75	1.00	0.75 0).75	1.00	0.89 0).70 2	1.00
	LRcc	0.16	0.23	0.82	0.16	0.23	0.61	0.13	0.05	0.01	0.07	0.00	0.28	0.38	0.05	0.28	0.38	0.05	0.28	0.02	0.09	0.61	0.28	0.30	0.61	0.97	0.64	0.61	0.90	0.95	0.61	0.88 0	33 0.6	0.90	0.95	0.61	0.90	0.95	0.61	0.90 0	J.95 (0.61	0.99 (J.82 (0.61
Nasdag - VXN	DOhit	0.08	0.57	0.95	0.08	0.57	1.00	0.32	0.48	0.80	0.05	0.31	0.90	0.46	0.48	0.90	0.46	0.48	0.90	0.35	0.95	1.00	0.68	0.90	1.00	0.83	0.85	1.00	0.76	0.75	1.00	0.96 0	90 1.0	0.76	0.75	1.00	0.76	0.75	1.00	0.76 (0.75	1.00	0.90 C	0.70	1.00
- moning + Mit	DOVoP	0.00	0.95	0.00	0.00	0.96	1.00	0.92	0.00	0.00	0.14	0.52	0.00	0.52	0.27	0.00	0.51	0.20	0.00	0.00	0.00	1.00	0.27	0.02	1.00	0.19	0.05	1.00	0.15	0.00	1.00	0.10 0	07 10	0.14	0.10	1.00	0.14	0.90	1.00	0.15 (0.00	1.00	0.96 (1.02	1.00
	DQvart	0.00	0.20	0.99	0.00	0.20	1.00	0.29	0.00	0.86	0.14	0.52	0.92	0.51	0.57	0.88	0.51	0.59	0.86	0.27	0.90	1.00	0.37	0.95	1.00	0.18	0.95	1.00	0.15	0.90	1.00	0.19 0	91 1.0	0.14	0.87	1.00	0.14	0.89	1.00	0.10 0	1.90	1.00	0.20 0	4.65	1.00
	MCuc	0.50	0.09	0.22	0.60	0.08	0.13	0.07	0.01	0.00	0.22	0.00	0.03	0.21	0.03	0.05	0.24	0.03	0.18	0.01	0.08	1.00	0.11	0.25	1.00	0.90	0.53	1.00	0.72	0.73	1.00	0.72 0	11 1.0	0.64	0.91	1.00	0.78	0.98	1.00	0.76 0	J.86	1.00	0.98 0).44	1.00
	MCind	0.27	0.08	0.21	0.28	0.09	0.00	0.25	0.04	0.67	0.35	0.22	0.70	0.19	0.10	0.11	0.18	0.10	0.11	0.84	0.50	0.00	0.84	0.78	0.00	0.67	0.19	1.00	0.75	0.96	1.00	0.48 0	56 1.0	0.74	0.96	1.00	0.74	0.96	1.00	0.74 0	J.96	1.00	0.49 C	J.38 ¹	1.00
	MCcc	0.56	0.17	0.41	0.57	0.17	0.00	0.51	0.08	0.65	0.71	0.45	0.60	0.37	0.19	0.24	0.36	0.18	0.23	0.32	0.99	1.00	0.31	0.44	1.00	0.68	0.37	1.00	0.52	0.09	1.00	0.97 0	85 1.0	0.52	0.09	1.00	0.52	0.09	1.00	0.51 (0.08	1.00	0.97 ().74	1.00
-		0.00			0.01	0.41		0.04		0.00	0.14	0.10	0.00	0.01	0.40	0.101	0.00	0.10	0.20	0.02	0.00	2.00	0.01		2.50	0.00	0.01	2.00	0.02	0.00	0	0.01	10		0.00	1.00	0.00	0.00		0.04 0			0.01		

Table 6: Results (in the form of p-values) for out-of-sample VaR accuracy for all GARCH, BEGE, univariate and bivariate Hawkes-POT models. Column headings denote the model at each level of significance α , the rows in each panel denote the test. The number of exceptions observed (Exc.) are also reported above the test results. All tests with a p-value less than 5% are shown in bold to highlight where the rejections of the accuracy tests are occurring.

		S&	P500 -	VIX	DA	X - VI	DAX	D	JI - V.	XD	Ni	kkei - V	VXJ	Nas	sdaq -	VXN
	α -level	0.95	0.99	0.999	0.95	0.99	0.999	0.95	0.99	0.999	0.95	0.99	0.999	0.95	0.99	0.999
	Model 1			*	*	*	*	*				*	*	*	*	*
Uni. Hawkes-POT	Model 2	*	*	*	*	*	*	*				*	*	*	*	*
	Model 3				*	*	*	*			*	*	*	*	*	
	Model 1	*	*	*	*	*	*	*	*	*	*	*	*			
Dir. Hamles DOT	Model 2	*	*	*	*	*	*	*		*	*	*	*	*	*	*
DIV. Hawkes-r O I	Model 3	*	*	*	*	*	*	*		*	*	*	*	*	*	*
	Model 4				*		*	*			*	*	*	*	*	
	GJRGARCH			*	*		*	*					*	*	*	
	GJRGARCH+IV			*	*		*	*					*	*	*	
	EGARCH						*	*					*	*	*	
Valatilita Madala	EGARCH+IV				*		*	*					*	*	*	
volatility models	GARCH						*	*					*	*	*	
	GARCH+IV						*	*					*	*	*	
	BEGE				*	*	*	*				*	*	*	*	
	BEGE+IV						*	*				*	*	*	*	*

Table 7: MCS results for comparing VaR forecast performance based on the asymmetric quantile loss function in equation 24, at each VaR level. The MCS results are based on a level of significance of $\alpha_M = 5\%$, with an * indicating that the model is a member of the final MCS.

		S&	P500 - V	IX	DA	AX - VDA	AX	I	DJI - VX	D	Ni	kkei - Vž	ζJ	Na	sdaq - V	XN
	α -level	0.95	0.99	0.999	0.95	0.99	0.999	0.95	0.99	0.999	0.95	0.99	0.999	0.95	0.99	0.999
VaR for $h =$	5															
	Model 1	31	1	0	30	1	0	34	1	0	35	2	0	34	5	0
		(0.22)	(0.03)	(0.32)	(0.33)	(0.03)	(0.32)	(0.07)	(0.03)	(0.32)	(0.05)	(0.13)	(0.32)	(0.05)	(0.94)	(0.33)
U.: Ules DOT	Model 2	29	2	0	31	2	0	33	1	0	38	1	0	36	5	0
Uni. Hawkes-PU1		(0.40)	(0.13)	(0.32)	(0.24)	(0.12)	(0.32)	(0.10)	(0.03)	(0.32)	(0.01)	(0.03)	(0.32)	(0.02)	(0.94)	(0.33)
	Model 3	9	0	0	23	1	0	27	2	0	32	2	0	26	3	0
		(0.00)	(0.00)	(0.32)	(0.66)	(0.03)	(0.32)	(0.65)	(0.13)	(0.32)	(0.15)	(0.13)	(0.32)	(0.70)	(0.37)	(0.33)
	Model 1	29	2	0	34	6	0	33	1	0	35	2	0	37	6	0
		(0.40)	(0.13)	(0.32)	(0.05)	(0.67)	(0.32)	(0.10)	(0.03)	(0.32)	(0.05)	(0.13)	(0.32)	(0.01)	(0.61)	(0.33)
	Model 2	29	2	0	34	6	0	33	2	0	35	2	0	36	5	0
Dia Harles DOT		(0.40)	(0.13)	(0.32)	(0.05)	(0.67)	(0.32)	(0.10)	(0.13)	(0.32)	(0.05)	(0.13)	(0.32)	(0.02)	(0.94)	(0.33)
BIV. Hawkes-PO1	Model 3	31	2	0	34	6	0	34	1	0	42	2	0	37	5	0
		(0.22)	(0.13)	(0.32)	(0.05)	(0.67)	(0.32)	(0.07)	(0.03)	(0.32)	(0.00)	(0.13)	(0.32)	(0.01)	(0.94)	(0.30)
	Model 4	16	2	0	23	2	0	27	2	0	38	3	0	27	3	0
		(0.05)	(0.13)	(0.32)	(0.66)	(0.12)	(0.32)	(0.65)	(0.13)	(0.32)	(0.01)	(0.34)	(0.32)	(0.56)	(0.37)	(0.33)
VaR for $h =$	10															
	Model 1	19	1	0	20	0	0	30	1	0	36	3	0	24	3	0
		(0.23)	(0.03)	(0.32)	(0.30)	(0.00)	(0.32)	(0.27)	(0.03)	(0.32)	(0.03)	(0.35)	(0.32)	(0.98)	(0.38)	(0.33)
Uni Hamkas DOT	Model 2	19	1	0	21	0	0	30	1	0	37	3	0	24	3	0
UIII. Hawkes-FUI		(0.23)	(0.03)	(0.32)	(0.42)	(0.00)	(0.32)	(0.27)	(0.03)	(0.32)	(0.02)	(0.35)	(0.32)	(0.98)	(0.38)	(0.33)
	Model 3	8	1	0	15	0	0	24	2	0	36	0	0	18	2	0
		(0.00)	(0.03)	(0.32)	(0.03)	(0.00)	(0.32)	(0.92)	(0.13)	(0.32)	(0.03)	(0.00)	(0.32)	(0.2)	(0.15)	(0.33)
	Model 1	18	1	0	31	1	0	30	1	0	36	3	0	23	5	0
		(0.15)	(0.03)	(0.32)	(0.22)	(0.03)	(0.32)	(0.27)	(0.03)	(0.32)	(0.03)	(0.35)	(0.32)	(0.85)	(0.92)	(0.33)
	Model 2	18	1	0	31	1	0	32	2	0	36	3	0	24	3	0
Bir Hawkoo POT		(0.15)	(0.03)	(0.32)	(0.22)	(0.03)	(0.32)	(0.14)	(0.13)	(0.32)	(0.03)	(0.35)	(0.32)	(0.98)	(0.38)	(0.33)
DIV. Hawkes-1 O I	Model 3	8	1	0	31	1	0	33	1	0	41	6	0	24	3	0
		(0.00)	(0.03)	(0.32)	(0.22)	(0.03)	(0.32)	(0.09)	(0.03)	(0.32)	(0.00)	(0.63)	(0.32)	(0.98)	(0.38)	(0.33)
	Model 4	12	1	0	16	1	0	25	2	0	40	2	0	18	2	0
		(0.00)	(0.03)	(0.32)	(0.05)	(0.03)	(0.32)	(0.92)	(0.13)	(0.32)	(0.00)	(0.14)	(0.32)	(0.20)	(0.15)	(0.33)
Expected		25.10	5.02	0.50	25.35	5.07	0.51	25.00	5.00	0.50	25.00	5.00	0.50	24.40	4.88	0.49
Shape $\hat{\xi}_u$			0.476			0.436			0.468			0.375			0.388	

Table 8: Results (in the form of p-values) for the LRuc test of VaR adequacy based on h = 5 and h = 10 day ahead forecasts. All tests with a p-value less than 5% are shown in bold to highlight where the rejections of the accuracy tests are occurring.